

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

MAR | 1989

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE OSWER Directive # 9833.3A

MEMORANDUM

SUBJECT: Interim Guidance on Administrative Records for

Selection of CERCLA Response Actions

FROM: J. Winston Porter

Assistant Administrator

TO: Regional Administrators, Regions I-X

This memorandum sets forth the policy and procedures governing administrative records for selection of response actions under the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA), as amended by the Superfund Amendments and Reauthorization Act of 1986 (SARA).

Attachment

cc: Director, Waste Management Division, Regions I, IV, V, VII, and VIII

Director, Emergency and Remedial Response Division,

Region II

Director, Hazardous Waste Management Division,

Regions III and VI

Director, Toxics and Waste Management Division, Region IX

Director, Hazardous Waste Division, Region X

Regional Counsel, Regions I-X

INTERIM GUIDANCE ON ADMINISTRATIVE RECORDS FOR THE SELECTION OF CERCLA RESPONSE ACTIONS

U.S. Environmental Protection Agency Office of Solid Waste and Emergency Response Washington, D.C. 20460

TABLE OF CONTENTS

I.,	INTR	ODUCTION													
	A.	Purpose and Scope of the Administrative Record													
	B.	Judicial Review													
	c.	Public Participation													
II.	PROCEDURES FOR ESTABLISHING THE ADMINISTRATIVE RECORD														
	A.	Administrative Record Coordinator													
	в.	Multiple Response Actions													
	c.	Compilation													
	D.	Index													
	E.	Location													
		1. General													
		2. Special Documents													
	F.	Public Availability													
		1. General													
		2. Remedial Actions													
		3. Removal Action													
	G.	Maintaining the Record													
	н.	Confidential File													
	I.	Copying													
	J.	Micrographics													
	к.	Certification													
III.	CONTENTS OF THE ADMINISTRATIVE RECORD														
	A.	Remedial Actions													
	B.	Removal Actions													
	c.	Imminent and Substantial Endangerment 26													
	D.	Public Comments													
	E.	Enforcement Actions													
		1. Negotiation Documents													
		2. PRP Lead RI/FS													
		3. Administrative Orders and Consent Decrees 28													
	F.	Excluded Documents													

	G.	Draf	t Docum	ments	and	Inte	rnal	. Me	mor	and	la	•	•	•	•		•	•	29
	H.	Priv	ileged	Docum	ents			•		•	•		•	•	•	•	•	•	30
	I.	Guid	ance Do	cumen	ts .			•		•	•	•	•	•	•	•		•	32
:	J.	Tech	nical 1	Litera	ture			•		•		•	•	•	•	•	•	•	33
	K.	Lega	l Sour	ces .				•		•	• .	•	•	•	•	•	•	•	34
	L.	NPL	Rulemal	cing D	ocke	t In	form	ati	on	•	• •	•		•	•	•	•	•	34
,	M.	RCRA	Docume	ents .				•		•	•	•	•	•	•	•	•	•	34
	N.	Post	-Decis:	ion Do	cume	nts	• •	•	• •	•	•	•	•	•	•	•	• .	• .	35
IV.	INVO	LVEME	NT OF C	THER	PART	IES		•		•	•	•	•	•	•	•	•	•	37
	λ.	Stat	es					•		•	•	•	•	•	•	•	•		37
		1.	State :	Involv	emen	t in	Fed	lera	l-L	ead	s	it	es		•	•	•	•	37
		2.	Federa:	Invo	lvem	ent	in S	tat	:e-L	ead	is	it	es		•		•	•	37
	B.	Fede	ral Fac	ciliti	es .			•		•		•		•			•	•	38
	c.	ATSD	R					•		•	•	•				•	•	•	39
	D.	Natu	ral Res	source	s Tr	uste	es .	•										•	40
						;													
v.	DISC	LAIME	Ř				٠	•		•	•	•			•	•			41
•																			
VI.	FURT	HER I	NFORMA:	CION .				•		•		•			•			•.	41
GLOSSARY							٠.	•		•		•	•		•				42
APPE	NDICE	s				•													
Appe	ndix	A.	Section	ons 11	3(j)	-(k)	of	CEF	CLA	•	•	•			•				45
Appe	ndix	в.	Model	File	Stru	ctur	e.	•		•			•	•	•	•	•	•	47
Appe	ndix	c.	Model	Index				•		•			•	•	•		•		51
Appe	ndix	D.	Model				ript	ion	fo	r A	dm	in	is	tra	at.	įv	e		
			Record				• •	•	• •	•	•	•	•	•	•	•			53
•	ndix		Comper								G	ui	daı	nce	9	• .	•	•	55
	ndix		Model								•						•	•	59
	ndix		Model																60
	ndix		Model																
• •	ndix		Model								-								62
Appe	ndix	J.	Micros	form A	ppro	val 1	Memo	ran	dum	•	•	•	•	• •	•	•	•	•	63
Appe	ndix	K.	Model	Certi	fica	tion		•		•	•	•	• •				•	•	64
Appe	ndix	L.	Preamb			_		•	•		_								
Appe	ndix	м.	Subpar	t I o	f the	e pro	pos	ed	NCP										73

#9833.3A

I. INTRODUCTION

A. Purpose and Scope of the Administrative Record

This guidance addresses the establishment of administrative records under §113 of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA), as amended by the Superfund Amendments and Reauthorization Act of 1986 (SARA). Section 113(k)(1) of CERCLA requires the establishment of administrative records upon which the President shall base the selection of a response action (see Appendix A for the complete statutory language).

Chapter I of this guidance introduces the purpose and scope of the administrative record. Chapter II reviews procedures for compiling and maintaining the administrative record. Chapter III examines the various types of documents which should be included in the administrative record. Chapter IV discusses the involvement in establishing the record of parties outside the Agency. Finally, this guidance includes a glossary of frequently used terms and acronyms as well as several appendices.

Although written for use by the United States Environmental Protection Agency (EPA), this guidance can be adapted for use by State and Federal agencies required to establish administrative records for the selection of CERCLA response actions. This guidance is not intended as a treatise on general principles of administrative law with regard to administrative records. This guidance reflects, to the extent feasible, proposed revisions to the National Contingency Plan (NCP) published on December 21, 1988 (53 FR 51394) (see Appendices L and M). The Agency may refine or change this memorandum when it promulgates the regulations in final form.

The administrative record established under §113(k) of CERCLA serves two primary purposes. First, the basis for the response selection is set forth in the record and under §113(j), judicial review of any issue concerning the adequacy of a response selection is limited to the record. Second, §113(k) requires that the administrative record act as a vehicle for public participation in the selection of the response action. This guidance document discusses procedures developed to ensure that the Agency's administrative records meet these twin purposes.

The basis of the Agency's selection of a response action must be documented in the administrative record. The record should include the documents considered or relied on by the Agency in making the response action decision for the site. For

^{*}References made to CERCLA throughout this memorandum should be interpreted as meaning "CERCLA, as amended by SARA."

purposes of this guidance, documents considered by the Agency are documents that were relied upon or comments which the Agency solicited or received from interested parties and the public. The following principles should be applied in establishing administrative records:

- The record should be compiled as documents relating to the selection of the response action are generated or received by the Agency;
- The record should include documents that form the basis for the Agency's decision, whether or not they support the Agency's response selection; and
- o The record should be a contemporaneous explanation of the basis for the selection of a response action.

The effort to establish adequate administrative records encompasses a vast array of people including: Administrative Record Coordinators, Remedial Project Managers (RPMs), On-Scene Coordinators (OSCs), enforcement staff, records management staff, Regional Counsel, Community Relations Coordinators (CRCs), other Federal agencies, States, CERCLA contractors, and the public. The roles and responsibilities of these people and their interaction forms an integral part of this guidance.

B. Judicial Review

Section 113(j)(1) of CERCLA provides that judicial review of any issues concerning the adequacy of any response action shall be limited to the administrative record.

Judicial review on the basis of an administrative record can provide numerous benefits. Under §113(j) of CERCLA and general principles of administrative law, the trial court reviewing the Agency's selection of response action is limited to the facts as set out in the administrative record. As a result, arguments that potentially responsible parties (PRPs) or citizens raise for the first time in court concerning the selection of a response action should not be considered.

Record review saves time by limiting the scope of trials, thereby saving the Agency's resources for cleanup rather than litigation. Courts will not allow a party challenging an Agency decision to use discovery, hearings, or additional fact finding to look beyond the Agency's administrative record, except in very limited circumstances. In particular, courts generally will not permit persons challenging the Agency's response decision to depose Agency decision-makers, staff, or contractors concerning the selection of the response action.

Furthermore, the administrative record may be cited long after officials responsible for the response decisions have moved

into different positions or have left the Agency. Judicial review limited to the record will save the Agency the time involved in locating former employees who may not remember the facts and circumstances underlying decisions made at a much earlier time.

Moreover, in ruling on challenges to the response action decision, the court will apply the highly deferential "arbitrary and capricious" standard of review set forth in §113(j)(2) of CERCLA. Under that standard, a court does not substitute its judgment for that of the Agency, but only overturns the response selection decision if it can be shown "on the administrative record, that the decision was arbitrary and capricious or otherwise not in accordance with the law."

However, the extent to which the Agency attains the benefits of judicial review limited to the record depends on the quality and completeness of each record. If the record is incomplete or fails to provide an adequate explanation of the basis for the Agency's response action selection, the court may remand the case to the Agency for reconsideration of its decision. Proper and thorough compilation of the record will help avoid these results.

C. Public Participation

Section 113(k)(2) of CERCLA requires that the public have the opportunity to participate in the development of the administrative record for response selection. Section 117 of CERCLA also includes provisions for public participation in the remedial action selection process. Both sections reflect a statutory emphasis on public participation in the development of the record and the response selection. Participation by interested persons will ensure that the Agency has considered the concerns of the public, including PRPs, in selecting a response action. In addition, for purposes of administrative and judicial review, the record will contain documents that reflect the views of the public, including PRPs, concerning the selection of a response action.

Without an opportunity for involvement of interested parties in the development of the administrative record, persons challenging a response action may argue that judicial review should not be limited to the record. The Agency must, therefore, make the information considered or relied on in selecting a response action available to the public, provide an appropriate opportunity for public comment on this information, place comments and information received from the public in the record, and reflect in the record the Agency's consideration of this information.

II. PROCEDURES FOR ESTABLISHING THE ADMINISTRATIVE RECORD

A. Administrative Record Coordinator

Each Region should designate a person to be the Administrative Record Coordinator. As delegated, the Record Coordinator generally has the duty of ensuring the adequate compilation and maintenance of the administrative record files. The Record Coordinator will not be responsible for deciding which documents are included in a record file. Those decisions should be made jointly by the OSC or RPM and the ORC. The Record Coordinator's duties ordinarily include:

- o Developing procedures for the creation of record files;
- o Ensuring notification of the availability of the record files for public inspection;
- o Ensuring availability of the record at the location at or near the site;
- o Coordinating efforts to obtain the necessary documents;
- O Updating the record files and indices on a regular basis (e.g., quarterly);
- o Compiling information in a central Regional location;
- o Indexing the record files;
- o Ensuring availability of the record file for copying;
- o Ensuring availability of validated sampling data, possibly at a location other than that of the record files:
- o Coordinating with ORC on questions of relevance and confidentiality of documents submitted for inclusion in the record files;
- o Maintaining the confidential portion of the record files, if necessary;
- o Maintaining the Regional compendium of guidance documents and technical references;
- o Coordinating with States and Federal agencies on record files compiled by them; and
- o Notifying appropriate personnel of the timing for review of State and Federal record files.

Appendix D contains a model position description for an Administrative Record Coordinator.

The Record Coordinator must work closely with RPMs, OSCs, enforcement staff, records management staff, Regional counsel, community relations staff, and the Department of Justice (DOJ) (for cases in litigation).

In the event that procedures associated with the compilation and maintenance of the record are drawn into question in litigation, the Record Coordinator may be called upon to prepare an affidavit or testify about those procedures. Therefore, the Record Coordinator should be familiar with the procedures associated with the record, and be qualified to fulfill the responsibilities outlined above.

B. Multiple Response Actions

Cleanups conducted under CERCLA are often broken up into distinct decision points within a total site response action. For remedial actions, these separate decision points are called "operable units." The response may be an action that fully addresses an operable unit, an "interim action," or a removal action taken before, during or after remedial actions.

In general, a separate administrative record forms the basis for each of these distinct actions. Every Record of Decision (ROD) or Action Memorandum must be supported by an administrative record. Information relevant to more than one response decision, such as a site inspection report or a preliminary assessment report may be placed in the record file for an initial response action and incorporated by reference in the indexes of subsequent record files for that site.

C. Compilation

The administrative record file should be compiled contemporaneously with the generation or receipt of pertinent information regarding a response selection. All documents for which there is no question as to the relevance and non-privileged nature of the information should be placed in the record file, entered into the index, and made available to the public as soon as possible after generation or receipt of the documents. For example, the remedial investigation/feasibility study (RI/FS) workplan, summaries of quality assured data, the RI/FS released for public comment, the proposed plan, and any public comments received on the RI/FS and proposed plan should be placed in the record file as soon as they are generated or received.

When there are questions as to whether particular documents should be included in the record file, such documents can be segregated and reviewed at regular intervals (e.g., every 3-4 months). For example, draft documents or documents subject to claims of privilege may be set aside for review by ORC and appropriate staff. At critical times, such as prior to the

public comment period, the issues regarding these documents should be completely resolved and the documents included in the record file if appropriate.

The record file may be updated while it is available for public inspection. The additional documents should be placed in the record file and entered in the index. It is not necessary to hold a public comment period for each update. Any updates to the record file should be made to all copies of the record file to maintain consistency.

All documentation of the selection of the response action should be in the record file when a decision document (i.e., ROD or Action memorandum) is signed. Documents relevant to the response selection but generated or received after the decision document is signed should be placed in a post-document decision file and may be added to the administrative record file in certain circumstances (see section III.N.).

D. Index

Each administrative record file must be indexed. The index plays a key role in enabling both agency staff and members of the public to help locate and retrieve documents included in the record file. In addition, the index can be used for public information purposes, identifying documents located elsewhere, e.g., included in the compendium of guidance documents (see Appendix E). The index also provides a concise overview of the history of the response action at the site.

The index also provides the Agency with a degree of control over documents located at or near the site. The creation of an index will prevent persons from altering the record simply by physically adding or removing documents from the file.

The index should include the following information for each document:

- o Document Number
- Document Title one or two line identification. Identify the actual document, not a transmittal memo or other less relevant document. Include sufficient information so the document cannot be confused with another (e.g., the title "report" may be insufficient).
- o Author Name and affiliation
- o Recipient Name and affiliation
- o Document Date date the document was finalized (or made available if never finalized), not the transmittal date or the receipt date.

o Document Location

The index can be organized either by subject or in chronological order. Appendix C contains a model index organized by subject. Computer databases have been helpful in generating and updating the index.

The index should be updated when the record file is updated. Such updates should coincide with the periodic updating of the record file and review of material for which there questions about relevance or privilege (see section II.C.). The index should be considered a "draft" index until the decision document is signed, at which time the index for the record file should be up-to-date.

E. Location

E.1. General

Section 113(k)(1) of CERCLA requires that the administrative record be available to the public "at or near the facility at issue." Duplicates of the record file may be kept at any other location.

A copy of the record file should be available for public inspection at the central Regional location. This copy of the record file should be maintained in a docket room and available for public inspection at reasonable times (e.g., 9-4, Monday-Friday). A copy of these files, with some exceptions described below, must be located at or near the site, as required by statute. In the case of an emergency removal, however, the record file need be available for public inspection only at the central location (see section II.F.3.).

The record file located at or near the site should be placed in one of the information repositories which may already exist for community relations purposes. These are typically located in a library, town hall, or other publicly accessible place. If there is no existing information repository, or if the repository does not have sufficient space for the record file, any other publicly accessible place may be chosen to house the record file. The Community Relations Coordinator (CRC) should be consulted on the location of the information repository and record file.

The record file should be transmitted to the local repository in coordination with the CRC. The CRC should make the initial contact to establish the local repository and request housing for the record file. The Record Coordinator should make arrangements for delivering the record file to the local repository.

The record file should include an introductory cover letter addressed to the librarian or repository manager (see Appendix F). In addition, a transmittal acknowledgement form should be included to ensure receipt of the record file (see Appendix G). Finally, an administrative record fact sheet should accompany the record to answer questions from the public (see Appendix H). Updates to the record file should be handled in a similar fashion (see section II.C.).

If feasible, a master copy of the record file should be kept at the Regional office or the central office of the lead agency. The master copy of the record file should not be accessible to the public in order to preserve the integrity of the record file. The master copy of the record file may be maintained in microform to conserve storage space (see section II.J.). If not feasible, the Region will need to establish an effective security system for the record file.

E.2. Special Documents

Certain documents which are included in the record file do not have to be maintained at or near the site or, in some cases, at the central Regional location, because of the nature of the documents and the burden associated with maintaining such documents in multiple locations. Such documents must, however, be incorporated in the record file by reference (i.e., in the index but not physically in the record file), and the index must indicate where the documents are publicly accessible.

Validated Sampling Data

Validated sampling data does not have to be located in the central Regional location or at or near the site. The sampling data may be left in its original storage location (e.g., Environmental Services Division (ESD) or contract laboratory). Data summary sheets, however, must be located in the file. The index must list the data summary sheets, reference the underlying validated sampling data, and indicate where that underlying data can be found.

Chain of Custody Forms

Chain of custody forms do not have to be located in the central Regional location or at or near the site. As with validated sampling data, chain of custody forms may be left in the original storage location. The index must reference the chain of custody forms and indicate their location.

Privileged Documents

When a privileged document is included in the record, it should be kept in a confidential portion of the record file. The confidential file should be kept in a locked cabinet at the

central Regional location. It should not be located at or near the site. The index should identify the title and location of the privileged document, and describe the basis for the asserted privilege. A short (non-privileged) description of the information in the privileged document should be inserted in the portion of the record file available to the public and included in the index. See also section III.H.

Guidance and Policy Documents

Guidance and policy documents may be kept in a compendium located in the Regional office to eliminate the need for reproducing copies of the same frequently used documents for each site record file. The documents included in such a compendium need not be physically included in the record file. The index must indicate which documents in the record file are located in the guidance compendium and specify the location of the guidance compendium. The index should reference the specific documents in the guidance compendium which were considered or relied on for the particular response action. See also section III.I. and Appendix E.

Technical Literature

Publicly available technical literature not generated for the site at issue (e.g., an engineering textbook), does not have to be located in the central Regional location or at or near the site. The document must be clearly referenced in the index. However, technical literature not publicly available must be physically included in the record file at the central location and at or near the site. See also section III.J.

F. Public Availability

F.1. General

Section 113(k) of CERCLA specifies that the administrative record "shall be available to the public." In satisfying this provision, the Agency must comply with all relevant public participation procedures outlined in §§113(k) and 117 of CERCLA; the proposed NCP (see Appendices L and M) contains additional guidance with respect to public availability (see also "Community Relations in Superfund: A Handbook," October 1988 - OSWER Directive No. 9230.0-3A; "Community Relations During Enforcement Activities," November 3, 1988 - OSWER Directive No. 9836.0-1A).

The availability of the record will vary depending upon the nature of the response action. Different procedures are outlined below for remedial and removal response actions.

In all cases, the Agency should publish a notice of availability of the record file when the record file is first made available for public inspection in the vicinity of the site

at issue. The notice should explain the purpose of the record file, its location and availability, and how the public may participate in its development.

The notice should be published in a major local newspaper of general circulation and distributed to persons on the community relations mailing list. This notice should also be sent to all known PRPs if they are not already included on the community relations mailing list. Publication of the notice should be coordinated with the Community Relations staff. A copy of the notice of availability should be included in the record file. Appendix I contains a model notice of availability.

Such a notice may be combined with other notices for the same site, such as a notice of availability of the community relations information repository, if they occur at the same time. Where desired, additional notice can be made in the <u>Federal Register</u>. In addition to the required newspaper notice, the public can be informed of the availability of the record file through existing Agency mechanisms, e.g., general and special notice letters, §104(e) information requests, and the community relations mailing list.

F.2. Remedial Actions

The administrative record file for a remedial action must be available for public inspection when the remedial investigation begins. For example, when the remedial investigation/feasibility study (RI/FS) work plan is approved, the Agency must place documents relevant to the selection of the remedy generated up to that point in the record file. Documents generally available at that time include the preliminary assessment (PA), the site investigation (SI), the RI workplan, inspection reports, sampling data, and the Community Relations Plan (CRP). The Agency must continue to add documents to the record file periodically after they are generated or received during the RI/FS process.

The record file must be publicly available both at a central Regional location and at or near the site at issue (see section II.E.). As mentioned above, a notice of the availability of the record file should be published in a major local newspaper when the record file is first made available for public inspection. In addition, the notice of availability should be sent to persons on the community relations mailing list.

With the completion of the RI/FS, the lead agency should undertake the following public participation procedures:

o Prepare a Proposed Plan which briefly analyzes the remedial alternatives studied in the RI/FS and proposes a preferred remedial action alternative;

- o Make the RI/FS report and Proposed Plan available at a central Regional location and at or near the site;
- o Publish in a newspaper a notice of availability and brief analysis of the RI/FS report and Proposed Plan;
- o Provide a formal comment period of not less than 30 calendar days for submission of comments on the Proposed Plan; [Note: the current NCP requires a public comment period of not less than 21 days. The Agency has proposed in the revised NCP a public comment period of not less than 30 days.]
- o Provide the opportunity for public meeting(s) in the affected area during the public comment period regarding the RI/FS and Proposed Plan;
- o Keep a transcript of the public meeting(s) on the RI/FS and Proposed Plan held during the comment period and make such transcript available to the public by including a copy in the record file;
- o Prepare a discussion (to accompany or be part of the decision document) of any significant changes to the Proposed Plan which occurred after the Proposed Plan was made available for public comment which are reflected in the ROD;
- o Prepare a response to each of the significant comments submitted during the public comment period to accompany the ROD (see section III.D.); and
- o Publish in a newspaper a notice of the availability of the ROD and make the ROD available to the public before commencement of any remedial action as required under §117(b) of CERCLA.

The Agency may hold additional public comment periods after signing the ROD. Comments received after signing the ROD should be placed in a post-decision document file and may be added to the record file in certain situations (see section III.N.).

F.3. Removal Actions

Section 113(k)(2)(A) of CERCLA requires that the Agency establish procedures for the appropriate participation of interested persons in the development of the administrative record for the selection of a removal action. "Appropriate" participation depends on the nature of the removal, as outlined below.

Non-Time-Critical Removal Actions

A non-time-critical removal action is a removal action for which, based on the site evaluation, the Agency determines that a planning period of at least six months exists before on-site removal activities must be initiated.

The administrative record file for a non-time-critical removal action must be made available for public inspection when the engineering evaluation/cost analysis (EE/CA) is made available for public comment. The record file must be available both at a central Regional location and at or near the site at issue (see section II.E.). A notice of the availability of the record file must be published in a major local newspaper and a copy of the notice included in the record file.

A public comment period on the EE/CA of not less than 30 days must be held so that interested persons may submit comments on the response selection for inclusion in the record file. A notice of the public comment period may be combined with the notice of availability of the record file if they occur at the same time. The Agency must respond to all significant comments received during the public comment period and place the comments and the responses to them in the record file (see section III.D.).

Comments generated or received after the decision document is signed related to selection of the removal action should be kept in a post-decision document file. They may be added to the record file in certain situations (see section III.N.).

Time-critical Removal Actions

A time-critical removal action is a removal action for which, based on the site evaluation, the Agency determines that a period of less than six months exists before on-site removal activities must be initiated. This category includes emergency removal actions which are described in greater detail below.

The administrative record file for these actions must be available for public inspection no later than 60 days after the initiation of on-site removal activity. If feasible, the record file should be made available earlier. The record file must be available both at a central Regional location and at or near the site at issue.

If, however, on-site cleanup activity is initiated within hours of the verification of a release or threat of a release and on-site cleanup activities cease within 30 days (emergency actions), the record file may be made available only at a central Regional location and need not be made available at or near the site.

For all time-critical removals, a notice of the availability of the record file must be published in a major local newspaper and a copy of the notice included in the record file.

A public comment period of not less than 30 days should be held in appropriate situations. In general, a public comment period will be considered appropriate if cleanup activity has not been completed at the time the record file is made available to the public and if public comments might have an impact on future action at the site. If a public comment period is considered appropriate, it should begin at the time the record file is made available for public inspection. The notice for the public comment period may be combined with the notice of availability of the record file if they occur at the same time. The Agency must respond to all significant comments received during the public comment period and place the comments and the responses to them in the record file (see section III.D.).

Whether or not the Agency holds a public comment period, comments received by the Agency related to the selection of the removal action must be placed in the record file. For information, including comments, generated or received after the decision document is signed - see section III.N.

G. Maintaining the Record

Document room procedures should be established to ensure orderly public access to the record files. In establishing public access procedures, the security and integrity of the record files must be maintained at all times.

The Agency prefers that each Regional Office have a reading area where visitors are able to review the record files. The record file must be available during reasonable hours (e.g., 9-4, Monday-Friday). The public reading area should include, wherever feasible:

- Administrative record files
- o Guidance Compendium (see section III.I.)
- o Access to a copier
- o Sign-in book

Controlled access to the files is accomplished by use of a visitor sign-in book. Sign-in books help minimize instances in which documents are lost or damaged. It also provides documentation of the Agency's efforts to provide public access to the record files. Pertinent information recorded in the book should include:

o Date of visit

- o Name
- o Affiliation
- o Address
- o Phone number
- o Site documents viewed
- o Cost of copied materials (if applicable)

Since documents in the record file should be complete, properly organized and legible, the integrity of the record file must be maintained. If possible, storage and reading areas should be supervised to maintain proper security. Documents should not leave the document room or be left unattended. To the extent feasible, the Record Coordinator should check the order of the documents after being viewed by the public to be certain all documents have been returned intact. The documents in the record file should be kept secure, either in a locked room or in locked cabinets.

The record file located at or near the site should be handled with similar care. If possible, the record file should be treated as a non-circulating reference; it should not leave the local repository except under supervision. The phone number of a record file contact should be provided to record file users and to the manager of the local repository so that problems can be identified and cured. Such information can be included in an informational fact sheet accompanying the record file (see Appendix H). In addition, the Record Coordinator should plan reviews of the local record files on a systematic basis.

Where the site is a fund-lead or PRP-lead, EPA should retain a master copy of the record file at a central Regional location. Where a State is the lead agency at a site, EPA should assure that the State maintains a master copy of the record file. The record files are permanent records that must be retained.

As to the local repository, the statute and proposed regulations are silent concerning the duration of public availability of the record file. The Agency's primary concern is public participation in selection of the response action. Following initiation of the response action, public interest in background information other than the Record of Decision, RI/FS or administrative record file may wane. In any event, the statutory provisions for judicial review and deadlines for filing cost recovery actions provide useful references for keeping the record file publicly available. See §§113(g) and (h) of CERCLA.

Where there is cost recovery litigation or such litigation is possible, the record file should be available at least until the completion of pending litigation concerning the site.

The record file continues to serve as a historical record of the response selection, even after the statute of limitations for cost recovery action has passed. Where there is considerable public interest, the local repository may wish to keep the record file available for public viewing.

H. Confidential File

In certain situations, documents in the record file may be subject to an applicable privilege (see section III.G.). Whenever feasible, information relevant to the response selection which is contained in a privileged document should be extracted in a releasable manner and included in the publicly accessible portion of the record file. The privileged document should be included in a confidential portion of the record file.

The Administrative Record Coordinator should maintain a confidential portion of the record file for such privileged documents. These documents should be listed in the index to the entire record file and identified as "privileged." The index should identify the title and location of the privileged document, and describe the basis for the asserted privilege. A short (non-privileged) description of the document and its privileged nature should, where possible, be placed in the files available to the public.

The confidential portion of the record file should be stored in locked files at a central Regional location and should not be located at or near the site. The confidential portion of the record file should be separate from the publicly available record file to protect against inadvertent disclosure. Each privileged document should be stamped "confidential" at the bottom of each page of the document. Where the material is not a written document (such as a computer disk or cassette tape) the jacket should be stamped "confidential." A complete list of all materials contained in the confidential portion of the record file should be maintained by the Record Coordinator. The coordinator should also maintain a log which will include the time, date, document name, and identity of persons checking out and returning materials to the confidential file.

A routine access list should be prepared for each record file or, in the future, as soon as a new record file is established. This routine access list must be approved by the Waste Management Division Director or the Environmental Services Division Director, and ORC. Once approval is given, persons on the list will be able to access the confidential files through the Record Coordinator. No one should have access to the

confidential files without the approval of the persons listed above.

This policy and procedure for privileged materials does not supersede any policy and procedures established by the Agency under the Freedom of Information Act (FOIA), 5 U.S.C., §552, and Agency regulations implementing FOIA at 40 C.F.R. Part 2. If the requester is in close proximity to the record file, the Agency may respond to FOIA requests for the administrative record by telling a requester the location and availability of the record file. Decisions regarding disclosures of materials under FOIA should be coordinated among the various agency officials with access to such materials.

I. Copying

Section 117(d) of CERCLA requires that each document developed, received, published, or made available to the public under §117 be made available for public inspection and copying at or near the site. Under §113(k)(2)(B) of CERCLA, these documents must also be included in the administrative record. Under these provisions of CERCLA, the lead agency must ensure that documents in the record files are available for copying but does not bear responsibility for copying the documents themselves. Therefore, it is preferable that the record file should be located in a facility which contains a copying machine (e.g., a public library).

Alternatively, the Agency may arrange for copying on behalf of a requester. The Agency may ask that requesters arrange for copying by contractors or commercial copy centers who then bill the requester directly.

The Agency should follow the FOIA regulations at 40 C.F.R. Part 2 in determining the appropriate charge for copying.

Generally, copying fees should be waived for other Federal Agencies, EPA contractors or grantees, and members of Congress. The Agency should charge \$.20 a page for paper copies as provided in 40 C.F.R. Part 2. Reproduction of photographs, microfilms or magnetic tapes, and computer printouts should be charged at the actual direct cost to the Agency.

J. Micrographics

The administrative record file may be maintained in microform such as microfilm or microfiche. Use of micrographics can significantly reduce the space required to store administrative record files. In addition, micrograpics can simplify the tasks of reproducing copies of the record and transmission of the record files to the local repositories. Any use of micrographics should be conducted in an orderly manner consistent with records management procedures. If using micrographics to maintain the record files, readers will need to

be available at the Regional office and with the record file located "at or near the site" to ensure public access to the record files.

Microform copies of original documents are admissible in court if created in an organized fashion. The Business Records as Evidence Act (28 U.S.C. §1732) specifies that copies of records, which are made "in the regular course of business" and copied by any process which accurately reproduces the original, are "as admissible in evidence as the original itself." See also Federal Rules of Evidence 1003. Thus, if the lead agency can demonstrate that CERCLA records and microform copies of those records are produced in the regular course of business, they would be admissible in court.

The Office of Information Resources Management (OIRM) has granted approval for the use of micrographics in establishing administrative records (see Appendix J). Any use of micrographics must still comply with the remaining provisions of Chapter 6 of the EPA Records Management Manual (7/13/84).

K. Certification

The Agency must certify the completeness of the administrative record when the record is filed in court. Appendix K contains a model court certification.

Such certification should be signed by the Regional Administrator's designee after consultation with ORC and DOJ. Any certification of the record should be made by program staff and not legal staff.

Although not required, the Region may choose to have the Administrative Record Coordinator certify that the record was compiled and maintained in accordance with applicable Agency regulations and guidance. Such certification would attest that the record was compiled in accordance with current Agency procedures and would not address the completeness of the record file.

III. CONTENTS OF THE ADMINISTRATIVE RECORD

A. Remedial Actions

The administrative record for selection of a remedial action should consist of:

- o documents which the Agency considered or relied on to select the remedial action; and
- o documents which demonstrate the public's opportunity to participate in and comment on the selection of the remedial action.

As stated earlier, the administrative record file for the selection of a remedial action must be made available for public inspection at the commencement of the remedial investigation phase. For example, when the RI/FS work plan is approved, the Agency must place documents on selection of remedy generated up to that point in the record file. Documents should continue to be added to the record file until a decision document is signed. Documents containing new information relevant to the response selection but generated or received after the decision document is signed should be kept in a post-decision document file and may be added to the record file in certain situations (see section III.N.).

Typically, but not in all cases, the record for a remedial response action will consist of the following types of documents. The list below is neither a statement of requirements for each record nor is it exclusive. The record file should include any other documents not listed below which meet the general criteria described in the first paragraph of this section.

Factual Information/Data

- o Preliminary Assessment (PA) report
- o Site Investigation (SI) report
- o Approved Remedial Investigation/Feasibility Study (RI/FS) workplan
- o Amendments to final workplan
- o Sampling and Analysis Plan (SAP): consisting of a quality assurance project plan (QAPP) and a field sampling plan (FSP)
- o Sampling data: validated data during the RI/FS, or any data collected for previous actions such as RCRA or removal actions which are considered or relied on in selecting the

remedial action. Unvalidated data should be included only if relied on by the Agency in the absence of validated data.

- o Chain of custody forms
- o Inspection reports
- o Fact sheet or summary information regarding remedial action alternatives generated when special notice letters are issued to PRPs at an early stage of the RI/FS (see "Interim Guidance on Notice Letters, Negotiations, and Information Exchange," October 19, 1987 OSWER Directive No. 9834.1)
- o Data summary sheets
- o Technical studies performed for the site (e.g., a ground water study)
- o Endangerment assessment/risk assessment and underlying documentation (see section III.C.)
- o RI/FS (as available for public comment and as final, if different)
- o Data submitted by the public, including PRPs

Policy and Guidance

- Memoranda on site-specific or issue-specific policy decisions. Examples include memoranda on off-site disposal availability, special coordination needs (e.g., dioxin), and applicable, or relevant and appropriate requirements (ARARs) (to the extent not in the RI/FS), cost effectiveness, and utilization of permanent solutions and alternative treatment technologies.
- o Guidance documents (see section III.I.)
- o Technical literature (see section III.J)

Public Participation

- o Community Relations Plan
- o Proposed Plan
- o Public notices: any public notices concerning response action selection such as notices of availability of information, notices of meetings, and notices of opportunities to comment
- o Documentation of public meetings: information generated or received during meetings with the public (including PRPs)

- and memoranda or notes summarizing significant information submitted during such meetings
- o Public comments: complete text of all written comments submitted (see also section III.D.)
- o Transcripts of public meetings: includes especially meetings held during the public comment period on RI/FS, Proposed Plan, and any waiver of ARARs under §121(d)(4) of CERCLA
- o Responses to significant comments: responses to significant comments received from the public concerning the selection of a remedial action
- o Responses to State and other Federal agency comments

<u>Enforcement Documents</u> (included if the document contains information relevant to the response selection or public participation in the response selection and does not pertain exclusively to liability)

- o Administrative orders
- o Consent decrees
- o Affidavits containing relevant factual information not contained elsewhere in the record
- o Notice letters to PRPs
- o Responses to Notice letters
- o §104(e) information request letters and subpoenas
- o Responses to §104(e) information request letters and subpoenas

Other Information

- o Index (see section II.D.)
- o Documentation of State involvement: documentation of the request and response on ARARS, §121(f)(1)(G) notices and responses, a statement of the State's position on the Proposed Plan (concurrence, nonconcurrence, or no comment at the time of publication), opportunity to concur in the selected remedy and be a party to a settlement (see section IV.A.)
- o health assessments, health studies, and public health advisories issued by the Agency for Toxic Substances and Disease Registry (ATSDR) (see section IV.C.)

o Natural Resource Trustee findings of fact and final reports and natural resource damage assessments (see section IV.D.)

Decision Documents

- o Record of Decision (ROD): final remedial action decision document (including responsiveness summary)
- o Explanations of significant differences (under §117(c)) and underlying information
- o Amended ROD and underlying information

The administrative record should both track the history of the site and be understandable to the reader. Appendix B provides a model file structure for organizing the record. Appendix C contains a model index.

B. Removal Actions

The administrative record for selection of a removal action should consist of:

- o documents which the Agency considered or relied on to select the removal action; and
- o documents which demonstrate the public's opportunity to participate in and comment on the selection of the removal action, when appropriate.

For a time-critical removal action, the record file should be available for public inspection as early as possible and must be available no later than 60 days after initiation of on-site removal activity. Documents prepared prior to the final decision document must be placed in the record file and made available to the public.

For a non-time-critical removal action, the record file must be made available for public inspection when the EE/CA is available for public inspection. Documents should continue to be added to the record file from the time the EE/CA is available until a decision document (i.e. Action memorandum) is signed.

For all removal actions, documents containing new information generated after the selection of a response action, other than comments made during the public comment period and responses to those comments, should be kept in a post-decision document file and may be added to the record file in certain situations (see section III.N.).

Typically, but not in all cases, the record for a removal response action will consist of the following types of documents.

The list below is neither a statement of requirements for each record nor is it exclusive. The record file should include any other documents not listed below which meet the general criteria described in the first paragraph of this section.

Factual Information/Data

- o Preliminary Assessment (PA) report
- o Site evaluation (SI) report
- o EE/CA (for a non-time-critical removal action)
- o Sampling plan
- o Sampling data: validated data obtained in support of the removal action, or any data collected for previous actions such as RCRA or other response actions which are considered or relied on in selecting the removal action. Unvalidated data should be included only if relied on by the Agency in the absence of validated data.
- o Chain of custody forms
- o Inspection reports
- o Technical studies performed for the site (e.g., a ground water study)
- o Endangerment assessment/risk assessment and underlying documentation
- o Data submitted by the public, including PRPs

Policy and Guidance

- Memoranda on site-specific or issue-specific policy decisions. Examples include memoranda on off-site disposal availability, compliance with other environmental statutes, special coordination needs, e.g., dioxin.
- o Guidance documents (see section III.I.)
- o Technical literature (see section III.J.)

Public Participation

- o Community Relations Plan
- Public notices: any public notices concerning response action selection such as notices of availability of information, notices of meetings, and notices of opportunities to comment

- o Documentation of public meetings: information generated or submitted during meetings with the public (including PRPs) and memoranda or notes summarizing significant information submitted during such meetings
- o Public comments: complete text of all written comments submitted (see also section III.D.)
- o Responses to significant comments: responses to significant comments received from the public concerning the selection of a removal action
- o Responses to State and other Federal agency comments

<u>Enforcement Documents</u> (included if the document contains information relevant to the response selection or public participation in the response selection and does not pertain exclusively to liability)

- o Administrative orders
- o Consent decrees
- o Affidavits containing relevant factual information not contained elsewhere in the record
- o Notice Letters to PRPs
- o Responses to Notice Letters
- o §104(e) information request letters and subpoenas
- o Responses to §104(e) information request letters and subpoenas

Other Information

- o Index (see section III.D.)
- o Documentation of State involvement (see section IV.A.)
- o ATSDR health assessments, health studies, and public health advisories (see section IV.C.)
- o Natural Resource Trustee findings of fact and final reports and natural resource damage assessments (see section IV.D.)

<u>Decision Documents</u>

- o EE/CA Approval Memorandum
- o Action Memorandum

o Amended Action Memorandum

The administrative record should both track the history of the site and be understandable to the reader. Appendix B provides a model file structure for organizing the record. Appendix C contains a model index.

C. Imminent and Substantial Endangerment

Under §106 of CERCLA, the Agency may find the existence of an imminent and substantial endangerment to the public health or welfare of the environment because of an actual or threatened release of a hazardous substance.

The determination of the existence of an imminent and substantial endangerment is an important component of the selection of the response action. Therefore, all documents considered or relied on in making that determination, including any endangerment assessment and its underlying documentation, must be included in the administrative record file. If the determination that there is an imminent and substantial endangerment is properly documented in the record file, judicial review of that determination in an action under §106 of CERCLA will also be limited to the administrative record.

D. Public Comments

In general, the administrative record should document any opportunity for public involvement in the selection of a response action. All documents related to the opportunity (e.g., notices and fact sheets), and relevant written comments received from the public should be included in the record file to establish a complete record for purposes of judicial review.

The record file should contain information brought to the Agency's attention by the public. Reports, data and other information generated by outside parties and submitted to the Agency should be included in the record file. Public requests for information (e.g., Freedom of Information Act (FOIA) requests for copies of reports), need not be included in the record file.

The Agency should request that substantive oral comments (either in person or over the phone) be put in writing by the commenter for inclusion in the record file. The commenter should be advised that the obligation to reduce the comment to writing rests with the commenter.

The Agency may respond to comments received prior to a public comment period in various ways, depending on the nature and relevance of a particular comment. The Agency's consideration of such a comment may be in the form of a written response, or reflected by documented actions taken after

receiving the comment, or even by changes in subsequent versions of documents. If the Agency prepares a written response to a comment, the response should be included in the record file.

The Agency may notify commenters that comments submitted prior to a formal public comment period must be resubmitted or specifically identified during the public comment period in order to receive formal response by the Agency. Alternatively, the Agency may notify a commenter that the Agency will respond to the comment in a responsiveness summary prepared at a later date. The Agency, however, has no duty to respond to any comments received before the formal public comment period, or to respond to comments during the public comment period until the close of the public comment period. The Agency may, in its discretion, respond in writing to comments submitted outside of a formal public comment period.

Comments received during the formal public comment period must be addressed in the responsiveness summary (included with the ROD in remedial response actions). The responses may be combined by subject or other category in the record file. Comments should be included in the record file in their original form wherever feasible.

Comments which are received after the formal comment period closes and before the decision document is signed should be included in the record file but labeled "late comment." Such comments should be handled as post-decision information (see section III.N.).

Comments received after the decision document is signed should be placed in a post-decision document file. They may be added to the record file in limited circumstances (see section III.N.).

E. Enforcement Actions

The same procedures should be used for establishing an administrative record whether or not a response action is selected in the context of an enforcement action. The following additional information, however, may assist the establishment of administrative records for selection of CERCLA response actions where there is enforcement activity.

E.1. Negotiation Documents

During negotiations with the Agency, a potentially responsible party (PRP) may produce documents and claim that they constitute confidential business information (CBI) or offers of settlement subject to Rule 408 of the Federal Rules of Evidence.

Generally, those documents are not part of the administrative record for response selection unless they are

submitted by PRPs for consideration in selecting a response action and are considered or relied on in selecting the response action. A privileged document which was considered or relied on in selecting the response action should be placed in the confidential portion of the record file. Such a document should be summarized and the summary included in the publicly accessible portion of the record file (see section II.H.). If the information cannot be summarized in a disclosable manner, the information should be placed in the confidential portion of the record file only and listed in the index to the file.

E.2. PRP-lead RI/FS

Where a PRP is conducting the RI/FS, the PRP must submit all technical information on selection of the remedial action generated during the RI/FS to the Agency. Technical information includes sampling data, workplans, reports, and memoranda. The Agency, and not the PRP, will establish the administrative record (see "Interim Guidance on Potentially Responsible Party Participation in Remedial Investigations and Feasibility Studies," May 16, 1988 - OSWER Directive No. 9835.1a).

PRPs may be delegated responsibility for some record file maintenance activities, such as housing the files at or near the site. PRPs cannot, however, be responsible for decisions on what documents comprise the record, because of, among other things, the potential for a conflict of interest.

E.3. Administrative Orders and Consent Decrees

Final administrative orders and consent decrees issued prior to selection of the response action (e.g. ordering a PRP to conduct the RI/FS), should be included in the administrative record file. Administrative orders or consent decrees issued after the signing of the ROD or the action memorandum should not be included in the record file, unless the consent decree or administrative order meets the criteria for the inclusion of post-decision documents in the record file (see section III.N.). In such cases, the documents may be added to the record file in accordance with procedures for post-decision documents. Drafts of administrative orders and consent decrees should not be included in the record file.

Note that administrative orders must also be supported by an administrative record. These administrative records are not the same as administrative records for the selection of a response action as described in this guidance. Some of the documents, however, which form the basis for a response decision may also form the basis for an order or decree. For example, some of the same documents which were considered or relied on to make the response decision embodied in a ROD will likely form the basis of a de minimis settlement embodied in a consent order. In such a case, the record file which forms the basis for the consent order

can reference the documents contained in the record file for the response decision.

F. Excluded Documents

Certain documents should not be included in the administrative record because they are irrelevant to the selection of the response action. Documents should be excluded from the administrative record if the Agency did not consider or rely on them in selecting the response action.

Material beyond the scope of the record should be kept in separate files maintained at the Regional office. These files need not be made publicly available, although many of the documents in the files may be available to the public if requested under FOIA.

Examples of documents that are irrelevant to the decision on selecting a response action may include Hazard Ranking System (HRS) scoring packages, contractor work assignments, cost documentation (as opposed to cost effectiveness information), and National List (NPL) deletion information. If, however, these documents contain information that is considered or relied on in the response action selection and that is not contained elsewhere in the record file, then the documents should be included in the record file.

Information regarding PRP liability is generally not included in the record file for selection of the response action except to the extent such information (typically substance specific) is considered or relied on in selecting the response action. Documents relating to PRP liability, however, should be compiled and maintained in the Regional office so that they are available at the time of notice to PRPs or referral of any litigation.

G. Draft Documents and Internal Memoranda

In general, only final documents should be included in the administrative record file. The record file should not include preliminary documents such as drafts and internal memoranda. Such documents are excluded from the record file because drafts and internal memoranda are often revised or superseded by subsequent drafts and memoranda prior to the selection of the response action. The preliminary documents are, therefore, not in fact considered or relied on in making the response action decision.

Drafts (or portions of them) and internal memoranda should be included, however, in two instances. First, if a draft document or internal memorandum is the basis for a decision (e.g., the draft contains factual information not included in a final document, a final document does not exist, or did not exist when the decision was made), the Agency should place the draft document or internal memorandum in the record file.

Second, if a draft document or internal memorandum is circulated to an outside party who then submits comments which the decisionmaker considers or relies on when making a response action decision, relevant portions of the draft document or the memorandum and the comments on that document should be included in the record file.

Examples of internal memoranda and staff notes which should not be included in the record file are documents that express tentative opinions or recommendations of staff to other staff or management, or internal documents that evaluate alternative viewpoints.

Drafts and internal memoranda may also be subject to claims of privilege (see section III.H.).

H. Privileged Documents

Some documents in the administrative record file may be protected from public disclosure on the basis of an applicable privilege. Any documents which are considered or relied on in a response action selection, but withheld from the public portion of the record file based on privilege, must be placed in a confidential portion of the record file (see section II.H.).

If a document is excluded from the public portion of the record file based on privilege, the relevant information should, if feasible, be extracted and included in the public record file. This can often be accomplished by deleting or redacting the privileged information from the document.

The privileges discussed below may be asserted by the Agency with respect to documents that are considered or relied on in the selection of a response action. The head of the office responsible for developing the document in question should assert the privilege. In all cases, the official asserting a privilege should consult with ORC.

Public disclosure of a privileged document may result in waiver of the privilege, although the nature and extent of the waiver will depend on the privilege asserted and the circumstances of the disclosure. In light of the potential for waiver, it is important that Agency personnel not release potentially privileged documents to the public without consulting with ORC.

<u>Deliberative Process</u>

The deliberative process privilege applies to predecisional, deliberative communications that express opinions,

advice, and recommendations of staff to other staff or management. The privilege functions to encourage the honest and free expression of opinion, suggestions and ideas among those formulating policy for government agencies (see "Guidance for Assertion of Deliberative Process Privilege," 10/3/84).

In general, if a document contains factual information forming the basis for the selection of the response action, it should be included in the record file.

Use of the deliberative process privilege should be balanced with the statutory mandate of including the public in the response action selection process. The privilege should be asserted if disclosure of the document will have an inhibiting effect on frank and open discussion among government staff and decisionmakers. Documents should not be withheld solely because they would reveal flaws in the case or information embarrassing to the government. Specific procedures exist for assertion of the deliberative process privilege, which include consulting with ORC.

Confidential Business Information (CBI)

The Agency must withhold from the public record trade secrets and commercial and financial information that is subject to protection under 40 C.F.R. Part 2. However, §104(e)(7) of CERCLA greatly restricts the assertions of confidentiality claims by PRPs at CERCLA sites. The Agency should attempt to avoid using CBI in making response action decisions and can do so in most cases by using other information instead. Where the Agency must use CBI in making its decision, 40 C.F.R. Part 2 and §104(e)(7) of CERCLA will apply and such information should be placed in the confidential portion of the administrative record file.

Attorney Work Product

This exclusion applies to documents prepared in anticipation of possible litigation. The work product privilege covers all documents prepared by an attorney or under an attorney's supervision, including reports prepared by a consultant or program employee. Litigation need not have commenced but it must be reasonably contemplated. These documents generally relate to enforcement or defensibility of a decision and are not considered or relied on in selecting a response action. These documents should not, therefore, be in the administrative record file.

Attorney-Client Communication

The attorney-client privilege applies to confidential communications made in connection with securing or rendering legal advice. The privilege is limited to communications where there was an intention to keep the information confidential.

Personal Privacy

This exemption covers information about individuals in personnel, medical, and similar files, the disclosure of which would constitute a clearly unwarranted invasion of personal privacy. The records must pertain to an individual, and not a business, to be excluded from the public portion of the administrative record file under this exemption. Often, information subject to the protection under the personal privacy privilege can be redacted from the document and the redacted version can be placed in the public portion of the record file.

State Secrets

An agency is authorized to exclude from public scrutiny information which, if released, would harm national security or interfere with the government's ability to conduct foreign relations. This privilege could be particularly important where the PRP is a Federal agency or a contractor for a Federal agency. In the case of a Federal facility cleanup, an Inter-Agency Agreement should spell out procedures for asserting this privilege.

Confidential Informant

Statements obtained from witnesses who have been granted confidentiality may be privileged.

Information Exempted by Other Statutes

Information specifically exempted from disclosure by a Federal statute need not be part of the public record. The statute in question must leave no discretion as to the requirement that matters be withheld from the public or it must establish particular criteria for withholding or refer to particular types of matters to be withheld.

I. Guidance Documents

Guidance documents, or portions of guidance documents, that are considered or relied on in selecting a response action should be included in the administrative record file for that response action. Any guidance documents generated for the site for which the record file is being compiled should be physically included in the record file. Certain guidance documents, however, do not have to be kept in the record file. Guidance documents not generated for the particular site for which the record is being compiled may be kept in a compendium of guidance documents maintained at a central Regional location.

Each EPA Regional office should maintain a compendium of guidance documents which are frequently used in selecting

response actions. As with an administrative record file, the compendium of guidance documents must be available to the public, but only at the central Regional location. The Administrative Record Coordinator should maintain and update the compendium of guidance documents. If a guidance document maintained in the compendium is considered or relied on when making a response action decision, the index to the record file must list the document and indicate its location and availability. See also Appendix E.

If a guidance document is listed in a bibliography to a document included in the record file (e.g., listed in the bibliography to the RI/FS), it need not be listed again in the index to the record file. In this case, however, the index must state that documents listed as bibliographic sources might not be listed separately in the index.

If a guidance document which is not included in the guidance compendium is considered or relied on in selecting the response action, the document should be physically included in the record file.

J. Technical Literature

Technical literature generated for the site at issue should be physically included in the administrative record file for that site, whether or not it is publicly available.

Similarly, technical literature not specifically generated for the site at issue which is not publicly available should also be included in the site-specific record file. Such documents include technical journals and unpublished documents that are not available through the Library of Congress or not circulated to technical libraries.

Publicly available technical literature not generated for the site at issue, however, need not be located at or near the site or at a central Regional location if the documents are referenced in the index to the record file. These documents do not have to be physically included in the record file because they are already available to public. Copying such documents creates a significant burden to the Agency and copyright laws may pose additional barriers to such copying. Examples of publicly available technical literature include engineering manuals, ground-water monitoring or hydrogeology textbooks, ATSDR toxicological profiles, and articles from technical journals.

Computer models and technical databases need not be physically included in the record file but should be referenced in the index to the record file. Printouts or other documents produced from the models and databases should be physically included in the record file if such documents contain information

which was considered or relied on in selecting the response action.

K. Legal Sources

Copies of statutes and regulations cited in documents included in the administrative record need not be included in the record file if they are readily available to the public. For example, the NCP and other regulations are easily accessible since they are published in the <u>Federal Register</u> and the Code of Federal Regulations.

Copies of the actual standards (statutes or regulations) comprising Federal and State ARARs should be physically included in the record file if they are not easily accessible. Also, other Federal and State criteria, advisories, and guidance documents pertinent to the site (i.e., what the Agency refers to as "TBCs," or standards "to be considered"), may not be easily accessible. If such documents are cited in an RI/FS, appendix to the RI/FS, EE/CA, or ROD, those advisories which are not readily available should be included in the record file.

L. NPL Rulemaking Docket Information

Generally, information included in the National Priorities List (NPL) rulemaking docket, such as the Hazard Ranking System (HRS) scoring package and comments received on the listing, need not be included in the administrative record for selection of a response action. The NPL docket contains information relevant to the decision to list a site, which may be irrelevant to the decision on response action selection.

Documents in the NPL docket which contain sampling data or other factual information which was considered or relied on in selecting a response action should be included in the record file if the information is not available already in the record file. Such information may include early sampling data taken by parties other than the Agency or its contractors (e.g., a State).

M. RCRA Documents

If an action is taken under CERCLA at a site with a history of Resource Conservation and Recovery Act (RCRA) activity, much of the information relating to those RCRA activities may be considered or relied on in making the CERCLA response action selection. Any relevant RCRA information, particularly information on waste management and RCRA corrective action at the site, should be included in the administrative record file (e.g., RCRA permit applications, inspection reports, RCRA Facility Assessment (RFA), RCRA Facility Investigation (RFI), Corrective Measures Studies (CMS), or responses to RCRA information requests).

Not all pre-existing RCRA information will be considered or relied on in selecting a CERCLA response action, but information on types of wastes, quantity of wastes, and observations of potential threats gathered during RCRA investigations should be included if not contained elsewhere in the record file.

N. Post-Decision Information

In general, documents generated or received by the Agency after signing the response action decision document (e.g., ROD or Action Memorandum), should not be added to the administrative record file. Since the administrative record contains the information which forms the basis for the response selection, documents generated or received after selecting the response action are not relevant to that response decision and should not be included in the record file. In limited situations, however, the Agency must include in the record file documents generated or received after the decision document is signed.

In all cases, documents generated or received after signing the decision document should be kept in a post-decision document file. This file is not part of the administrative record file and should be maintained at a central Regional location. Documents kept in the post-decision document file may be added to the record file in the situations described below:

- o Where a decision document does not address or reserves a significant aspect of a response action decision to be made at a later date. For example, a decision document that does not resolve the type of treatment technology. In such cases, the Agency should continue to add documents to the record file which form the basis for the unaddressed portion of the decision.
- o Where there is a significant change in the selected response action. Changes that result in a significant difference to a basic feature of the selected remedial action (e.g., timing, ARARs), with respect to scope, performance, or cost may be addressed in an explanation of significant differences. Under §117(c) of CERCLA,
 - "(a) fter adoption of a final remedial action plan (1) if any remedial action is taken, (2) if any enforcement action under section 106 is taken, or (3) if any settlement or consent decree under section 106 or section 122 is entered into, and if such action, settlement, or decree differs in any significant respects from the final plan, the President or the State shall publish an explanation of the significant differences and the reasons such changes were made."

The record file should include the explanation of significant differences, underlying documentation for the response action changes, any significant comments from the public, and Agency responses to any significant comments. A formal public comment period is not required on an explanation of significant differences.

Some changes will be so significant that they fundamentally alter the very nature or basis of the overall response action. Such changes will require an amended decision document. The Region will decide whether a change to a response action is considered a significant or a fundamental change for purposes of addressing the change (the "Guidance on Preparing Superfund Decision Documents: The Proposed Plan and Record of Decision," OSWER Directive No. 9355.3-02) is expected to address this.

When the decision document is amended, the amended decision document, the underlying documentation, any significant comments from the public, and Agency responses to any significant comments, should be included in the record file. ROD amendments will require a formal public comment period.

- Where comments containing significant information are 0 generated or received after the close of the public comment period. The Agency must consider comments submitted after the public comment period only to the extent that the comments contain significant information not contained elsewhere in the record file which could not have been submitted during the public comment period and which substantially support the need to consider significantly altering the response action. Documents meeting this test should be included in the record file, along with the Agency's responses to the significant comments. Whether or not such information results in a change to the selected decision. In this case, the comments, the underlying documentation, and Agency responses to such comments, should be included in the record file.
- Post-decision information may also be added to the record file if the Agency holds public comment periods after the selection of the response action. The Agency may hold additional public comment periods or extend the time for submission of public comment on any issue concerning response selection. Such comment should be limited to the issues for which the Agency requested additional comment. All significant comments submitted during such comment periods, along with any public notices of the comment period, transcripts of public meetings, and Agency responses to the comments, should be placed in the record file.

IV. INVOLVEMENT OF OTHER PARTIES

A. States

A.1. State Involvement on Federal-Lead Sites

The administrative record for a Federal-lead site must reflect the opportunity for involvement in the selection of the response action afforded the particular State. The record for a remedial action should include documents that reflect at least the following State participation or the opportunity for State participation:

- o Letter to State requesting identification of ARARs and the final response from State identifying ARARs (and certification from the State);
- o Opportunity to comment or comments on a proposed finding or decision to select a remedial action not attaining a level or standard of control at least equivalent to a legally applicable or relevant and appropriate standard, requirement, criteria, or limitation (as mandated by §121(d)(4) of CERCLA);
- Opportunity to comment or comments on the final draft RI/FS, the Proposed Plan and EPA responses to the comments;
- o Significant post-decision State comments and EPA responses to the comments (place in the post-decision document file for possible inclusion in the record file see section III.N.).

The administrative record for a removal action should reflect any State participation, especially any State comments and EPA responses to the comments.

The record file should only include final State comments. Any preliminary deliberations between the State and EPA relevant to the response selection need not be part of the record file if superseded by documentation of the State's final position.

The governing body of an Indian tribe should be afforded the same treatment as a State in accordance with §126 of CERCLA.

A.2. Federal Involvement in State-Lead Sites

Where a State has been officially designated the lead agency for a CERCLA site, the State must compile the administrative record for that site in accordance with §113(k) of CERCLA. Since EPA has ultimate responsibility for both the selection of a response action (EPA signs the ROD) and the record on which that response action is based, EPA must participate in compiling and maintaining the record. In such cases, EPA must assure that the

record file forms a complete basis for the selection of the response action.

The State as lead agency must maintain the record file at a State office (e.g., the State's central environmental agency office) and at or near the site. At a minimum, the State as lead agency also must transmit a copy of the index, RI/FS workplan, the RI/FS released for public comment, the Proposed Plan, and any public comments received on the RI/FS and the Proposed Plan to the appropriate EPA Regional office. These documents should be transmitted to the Agency as they are generated or received. Transmittal of the index will not suffice. In addition, other documents may be requested by EPA on a case-by-case basis.

The Superfund Memorandum of Agreement (SMOA), or Cooperative Agreement (CA), must address the administrative record requirements. The following language should be included in the SMOA or CA where the State has been officially designated the lead agency for a CERCLA site:

"The State must compile and maintain the administrative record upon which the selection of the [remedial, removal] action is based. The compilation and maintenance of the record must be, to the extent feasible, in accordance with EPA guidance on the administrative record. The administrative record must be located at the State [environmental agency] office, and at or near the site. In addition, the State must submit copies of the index, the RI/FS workplan, the RI/FS released for public comment, the Proposed Plan, and any public comments received on the RI/FS and Proposed Plan to the EPA Regional office, as they are added to the administrative record file. In addition, other documents may be requested by EPA on a case-bycase basis. The State shall comply with §113 of CERCLA and any applicable regulations. EPA may require the retention of other documents for cost recovery purposes."

The record file compiled by the State should reflect EPA's participation, comments, concurrence, and disagreements at the same stages as are required for State involvement in a Federal-lead site. The State must place in the record file any documents submitted by EPA for inclusion in the record file.

B. Federal Facilities

Federal agencies have the responsibility, pursuant to Executive Order 12580, to establish the administrative record for Federal facilities under their jurisdiction, custody, or control where using CERCLA authority for a response action. The record file for a Federal facility must include all documents considered or relied on in selecting a response action, including documents

submitted by EPA on the selection of the response action. The Federal agency must comply with all NCP (see Appendix M) and CERCLA requirements in compiling and maintaining the record, including the minimum public participation requirements in §§113 and 117 of CERCLA.

The Federal agency must maintain the record file at or near the site and ensure easy public access to the record file. If, for example, a site is a Department of Defense facility, the record file should be housed in a location which does not require military clearance for access. The Federal agency should keep a complete copy of the record file at a location within the Federal agency office comparable to an EPA Regional office.

At NPL sites and any other site where EPA is involved in selecting a response action at a Federal facility, EPA must participate in compiling and maintaining the record. In such cases, EPA must assure that the record file forms a complete basis for the selection of the response action. At a minimum, the Federal agency must transmit a copy of the index, the RI/FS workplan, the RI/FS released for public comment, the Proposed Plan, and any public comments received on the RI/FS and Proposed Plan to the appropriate EPA Regional office. These documents should be transmitted to the Agency as they are generated. Transmittal of the index will not suffice. In addition, other documents may be requested by EPA on a case-by-case basis. Inter-Agency Agreements (IAGs) should spell out procedures for compiling and maintaining the record.

C. ATSDR

Participation in the selection of a response action by the Agency for Toxic Substance and Disease Registry (ATSDR) should be reflected in the administrative record. The record file must include the initial and subsequent health assessments and any other information EPA solicits and obtains from ATSDR which EPA considers or relies on in its selection of a response action.

Draft versions of the health assessment and other draft documents upon which ATSDR comments should not be included in the record file. If, however, EPA solicits comments from ATSDR on a draft document such as a draft work plan or RI report, and receives formal comments from ATSDR which EPA considers or relies on in selecting a response action, then the document and comments should be included in the record file.

In the event that the ATSDR health assessment and EPA's risk assessment appear inconsistent, a document explaining the difference should be generated and placed in the record file.

D. Natural Resources Trustees

Section 122(j)(1) of CERCLA requires that the Agency give notice to the Natural Resources Trustee of a release or threatened release of any hazardous substance which may have resulted in damages to natural resources. The administrative record file must include the Agency's notice to the Natural Resources Trustee, and any subsequent final communications (e.g., a release or final report). In addition, any factual information provided by the Natural Resources Trustee which the Agency considers or relies on in selecting a response action should be included in the record file.

In the event that the Natural Resources Trustee's damage assessment and EPA's risk assessment appear inconsistent, a document explaining the difference should be generated and placed in the record file.

V. DISCLAIMER

The policies and procedures established in this document are intended solely for the guidance of employees of the U.S. Environmental Protection Agency. They are not intended and cannot be relied upon to create any rights, substantive or procedural, enforceable by any party in litigation with the United States. EPA reserves the right to act at variance with these policies and procedures and to change them at any time without public notice.

VI. FURTHER INFORMATION

For further information concerning this memorandum, please contact Sven-Erik Kaiser in the Office of Waste Programs Enforcement at FTS (202) 475-9806.

GLOSSARY

<u>Administrative Record</u>: as used in this guidance, the completed compilation of documents the Agency considered or relied on in selecting a response action.

Administrative Record File: as used in this guidance, the ongoing collection of documents the Agency anticipates will constitute the administrative record when the selection of response action is made.

<u>ARAR</u>: applicable or relevant and appropriate requirements (see §121(d) of CERCLA).

ATSDR: Agency for Toxic Substance and Disease Registry.

<u>CA</u>: cooperative agreement (entered into with a State or local government to transfer funds to conduct response activities).

CBI: confidential business information.

<u>CERCLA</u>: Comprehensive Environmental Response, Compensation, and Liability Act of 1980 as amended by the Superfund Amendments and Reauthorization Act of 1986 (also known as Superfund).

C.F.R.: Code of Federal Regulations.

CMS: corrective measure study (RCRA corrective action document,
equivalent to an FS).

CRC: Community Relations Coordinator.

CRP: community relations plan.

<u>Document</u>: as used in this guidance, includes writings, drawings, graphs, charts, photographs, and data compilation from which information can be obtained. It does not, however, include physical samples.

DOJ: Department of Justice.

<u>EE/CA</u>: engineering evaluation/cost analysis (removal document).

EPA: United States Environmental Protection Agency.

ESD: Environmental Services Division.

Explanation of Significant Differences: post-ROD document described in §117(c) of CERCLA.

FOIA: Freedom of Information Act.

FSP: field sampling plan.

HRS: Hazard Ranking System.

IAG: inter-agency agreement (made with a Federal agency).

<u>Lead Agency</u>: the agency that provides the OSC or RPM to plan and implement a response action under the NCP.

NCP: National Oil and Hazardous Substances Contingency Plan, proposed revisions published on December 21,1988 (53 FR 51394).

NPL: National Priorities List.

OECM: EPA Office of Enforcement and Compliance Monitoring.

OERR: EPA Office of Emergency and Remedial Response.

OIRM: EPA Office of Information Resources Management.

Operable Unit: a discrete action that comprises an incremental step toward comprehensively addressing site problems (see section 300.5 of the proposed NCP).

ORC: EPA Office of Regional Counsel.

OSC: On-Scene Coordinator (project manager for a removal action)

OSWER: EPA Office of Solid Waste and Emergency Response.

OWPE: EPA Office of Waste Programs Enforcement.

PA: preliminary assessment.

PRP: potentially responsible party.

QAPP: quality assurance project plan.

RA: remedial action.

<u>RCRA</u>: the Solid Waste Disposal Act as amended by the Resource Conservation and Recovery Act.

RD: remedial design.

RI/FS: remedial investigation/feasibility study.

RFA: RCRA facility assessment (RCRA document, equivalent to a PA/SI).

RFI: RCRA facility investigation (RCRA corrective action document, equivalent to an RI). <u>ROD</u>: Record of Decision (documents the selection of a remedial action).

<u>RPM</u>: remedial project manager (project manager for a remedial action).

SAP: sampling and analysis plan.

<u>SARA</u>: Superfund Amendments and Reauthorization Act of 1986 (see CERCLA above).

Site File: the file containing all site documentation.

SI: site investigation.

SMOA: Superfund memorandum of agreement (made with a State).

<u>Support Agency</u>: the agency that provides the support agency coordinator to furnish necessary data to the lead agency, review response data and documents, and provide other assistance as requested by the lead agency. The support agency may also concur on decision documents.

APPENDIX A

SECTION 113 (J) OF CERCLA

(i) JUDICIAL REVIEW .-

(1) LIMITATION.—In any judicial action under this Act, judicial review of any issues concerning the adequacy of any response action taken or ordered by the President shall be limited to the administrative record. Otherwise applicable principles of administrative law shall govern whether any supplemental materials may be considered by the court.

(2) STANDARD.—In considering objections raised in any judicial action under this Act, the court shall uphold the President's decision in selecting the response action unless the objecting party can demonstrate, on the administrative record, that the decision was arbitrary and capricious or otherwise not in

accordance with law.

(3) REMEDY.—If the court finds that the selection of the response action was arbitrary and capricious or otherwise not in accordance with law, the court shall award (A) only the response costs or damages that are not inconsistent with the national contingency plan, and (B) such other relief as is consistent with the National Contingency Plan.

(4) PROCEDURAL ERRORS.—In reviewing alleged procedural errors, the court may disallow costs or damages only if the errors were so serious and related to matters of such central relevance to the action that the action would have been signifi-

cantly changed had such errors not been made.

(h) Administrative Record and Participation Processius.—

(1) Administrative record.—The President chall citablish on edministrative record uson which the President chall been the exection of a reporte extent. The edministrative resident chall be evaluable to the public of or rest the facility of inner. The President also may place duplicates of the edministrative record of any other location.

(2) PARTICIPATION PROCEDURES.—

(A) REMOVAL ACTION.—The Project chell promulgate regulations in excerdance with chapter 5 of title 5 of the United States Code establishing procedures for the appropriate participation of interested persons in the descriptment of the administrative record on which the Projects will been the election of removal extions and on which ju-

dicial riview of rimoval actions will be beened

(B) REMEDIAL ACTION.—The Provident chall provide for the participation of intersed persons, including potentially responsible parties, in the development of the administrative record on which the Provident will been the exlection of remedial actions and on which judicial review of remedial actions will be based. The procedures developed under this subparagraph shall include, at a minimum, each of the following:

(i) Notice to potentially effected persons and the public, which shall be accompanied by a brief analysis of the plan and alternative plans that were considered.

(ii) A resonable opportunity to comment and provide

information regarding the plan

(iii) An opportunity for a public meeting in the affaced area in accordance with coesson 117(a)2) (relating to public participation).

(iv) A response to each of the significant comments, enticisms, and new data submitted in written or oral

presentations.

(v) A statement of the boow and purpose of the celect-

For purposes of this subparegraph, the edministrative record shall include all items developed and received under this subparegraph and all items described in the excond sentence of section 117(d). The President shall promulgate regulations in accordance with chapter 5 of title 5 of the United States Code to carry out the requirements of this

subparegraph.

(C) INTERIM RECORD.—Until such regulations under oubparagraphs (A) and (B) are promulgated, the administrative record shall consist of all teems developed and received
pursuant to current procedures for exlection of the response
action, including procedures for the participation of interacted parties and the public. The development of an admininterior record and the exlection of response action under
this Act shall not include an adjudicatory hearing.

(D) POTENTIALLY RESPONSIBLE PARTIES.—The President shall make reasonable efforts to identify and notify potentially responsible parties as early as possible before relection of a response action. Nothing in this paragraph shall

be construed to be a defence to liability.

APPENDIX B

MODEL FILE STRUCTURE

This model file structure may be used to compile an administrative record file for a remedial action, a removal action, or a combination of both remedial and removal actions. If the record documents a remedial action decision, section 2 of the file will contain only those removal action documents which (a) predate the remedial record of decision and (b) are relevant to the selection of the remedial action. If the record documents a removal action decision, sections 3, 4, and 5 of the file will contain only those remedial action documents which (a) predate the removal action memorandum and (b) are relevant to the selection of the removal action.

Justification is unnecessary for file categories without any documents. Those categories should be left out of the index.

A document should be filed in only one category, even if it falls into more than one category. It may be referenced in another category. If necessary, additional subcategories may be developed to accommodate documents not falling in any of the defined subcategories. Avoid adding categories of miscellaneous documents.

The correspondence subcategory can include comments and responses specific to the category. If the comments and responses are general in nature or address more than one category, they may be included in the public participation category.

INDEX [FIRST DOCUMENT]

1.0 SITE IDENTIFICATION

- 1.1 Background RCRA and other information
- 1.2 Notification/Site Inspection Reports
- 1.3 Preliminary Assessment (PA) Report
- 1.4 Site Investigation (SI) Report
- 1.5 Previous Operable Unit Information

2.0 REMOVAL RESPONSE

- 2.1 Sampling and Analysis Plans
- 2.2 Sampling and Analysis Data/Chain of Custody Forms
- 2.3 EE/CA Approval Memorandum (for non-time-critical removals)
- 2.4 EE/CA
- 2.5 Action Memorandum
- 2.6 Amendments to Action Memorandum

3.0 REMEDIAL INVESTIGATION (RI)

- 3.1 Sampling and Analysis Plan (SAP)
- 3.2 Sampling and Analysis Data/Chain of Custody Forms
- 3.3 Work Plan
- 3.4 RI Reports

4.0 FEASIBILITY STUDY (FS)

- 4.1 ARAR Determinations
- 4.2 FS Reports
- 4.3 Proposed Plan
- 4.4 Supplements and Revisions to the Proposed Plan

5.0 RECORD OF DECISION (ROD)

- 5.1 ROD
- 5.2 Amendments to ROD
- 5.3 Explanations of Significant Differences

6.0 STATE COORDINATION

- 6.1 Cooperative Agreements/SMOAs
- 6.2 State Certification of ARARs

7.0 ENFORCEMENT

- 7.1 Enforcement History
- 7.2 Endangerment Assessments
- 7.3 Administrative Orders
- 7.4 Consent Decrees
- 7.5 Affidavits
- 7.6 Documentation of Technical Discussions with PRPs on Response Actions
- 7.7 Notice Letters and Responses

8.0 HEALTH ASSESSMENTS

- 8.1 ATSDR Health Assessments
- 8.2 Toxicological Profiles

9.0 NATURAL RESOURCE TRUSTEES

- 9.1 Notices Issued
- 9.2 Findings of Fact
- 9.3 Reports

10.0 PUBLIC PARTICIPATION

- 10.1 Comments and Responses
- 10.2 Community Relations Plan
- 10.3 Public Notice(s) (Availability of the Administrative Record File, Availability the Proposed Plan, Public Meetings)
- 10.4 Public Meeting Transcripts
- 10.5 Documentation of Other Public Meetings
- 10.6 Fact Sheets and Press Releases
- 10.7 Responsiveness Summary
- 10.8 Late Comments

11.0 TECHNICAL SOURCES AND GUIDANCE DOCUMENTS

- 11.1 EPA Headquarters Guidance
- 11.2 EPA Regional Guidance
- 11.3 State Guidance
- 11.4 Technical Sources

APPENDIX C

MODEL INDEX

Attached is an excerpt of the Index of documents included in the Administrative Record for the Love Canal site. The Index lists the documents according to the EPA file structure (category number). The Index includes the following information fields:

DOCUMENT NUMBER	indicates the first and last page numbers of the document. Both page numbers will be the same for one-page documents. In this particular index, the document number consists of a three letter site code followed by microfilm reel and frame numbers.
TITLE	indicates the title or an enhanced description of the document in parentheses.
AUTHOR	indicates the author or primary originator and the author's corporate affiliation.
RECIPIENT	indicates the addressee or primary recipient and the addressee's corporate affiliation.
DATE	<pre>indicates document date by month/day/year. / / means no date was available.</pre>
TYPE	indicates the document type.
CATEGORY	indicates the EPA file structure number.

69/89/87

Administrative Record - Category Number Order LOVE CANAL

Page: 185

Document Number: LDV-001-1079 To 1089

Date: **02/07/86**

Title: Additional Sampling of the Black, Bergholtz & Cayuga Creeks and Extended Sewer Inspection

Project - Summary

Type: PLAN

Category: 11.4.0 Public Meeting Transcripts

Author: none: NY Dept of Environmental Conservation

Recipient: Garbarini, Douglas: US EPA

Partiment Number: LOV-801-1069 To 1078 Date: 11/13/86

Title: (Status report of Love Canal Remedial Programs for public meeting held on 11/13/85)

Type: GRAPHIC

Category: 11.4.0 Public Meeting Transcripts

Author: Bobersky, Guy T: NY Dept of Environmental Conservation

Recipient: none: none

Document Number: LCV-209-0013 To 0018 Date: 06/12/87

Title: (Love Canal Feasibility Study Workshop, schedule of days events and attendance sheets)

Type: CORRESPONDENCE

Category: 11.4.0 Public Meeting Transcripts

Condition: INCOMPLETE
Author: none: none
none: none

.

Document Number: LOV-009-0019 To 0024 Date: 08/25/87

Title: (Presentation of Proposed Plan for Love Canal)

Type: GRAPHIC

Category: 11.4.3 Public Meeting Transcripts

Condition: INCOMPLETE
Author: none: none
Recipient: none: none

APPENDIX D

MODEL POSITION DESCRIPTION FOR ADMINISTRATIVE RECORD COORDINATOR

INTRODUCTION

The incumbent serves as an Administrative Record Coordinator in one of the Regional offices of the Environmental Protection Agency (EPA). [Each Region may want to add an introduction to Superfund and the Regional office here.] The incumbent is responsible for compiling and maintaining administrative record files for CERCLA (Superfund) response action decisions.

Section 113(k) of CERCLA requires the establishment of an administrative record upon which the selection of a response action is based. Such a record is a compilation of all documents which the Agency considered or relied on in making its response action decision. Judicial review of any issues concerning the adequacy of any response action decision is limited to the administrative record. Public participation in the development of the record is required by law.

Establishment of thorough and complete administrative records is essential to EPA's Superfund program. Administrative records which include public participation and withstand judicial scrutiny allow EPA to meet its goals and objectives.

The incumbent will be responsible for compiling and maintaining administrative records for large numbers of Superfund sites. Each record requires coordination with many people including: Federal staff, State and local officials, private contractors, the general public and potentially responsible parties. Further responsibilities include deliberations over which materials to include in each record and requirements for dealing with privileged materials.

MAJOR DUTIES AND RESPONSIBILITIES

- 1. The incumbent is responsible for compiling and maintaining all of the administrative records for selection of CERCLA response actions for a Regional office of the EPA. The incumbent must have complete knowledge of all rules and procedures governing development of the administrative record files.
- 2. Receives and reviews all documents submitted by the Remedial Project Manager (RPM), On-Scene Coordinator (OSC), Office of Regional Counsel (ORC) and other appropriate staff for inclusion in the administrative record files. The incumbent will coordinate with staff responsible for deciding what

documents are included in the record and will arrange for adding documents to the record file.

- 3. Compiles the administrative record file for each CERCLA response action. This includes logging the receipt of each document, maintaining a central master file of documents, redacting information from privileged documents as directed by ORC, maintaining any privileged portions of each record using Agency security measures, arranging for copying of documents in each record and transmitting the documents to appropriate repositories.
- 4. Coordinates the compilation of the administrative record files with State and Federal agencies. This includes receiving records maintained by State and Federal agencies and notifying appropriate personnel of these records for their review.
- 5. Maintains and updates (monthly) an index of each administrative record file in conformance with Agency guidelines.
- 6. Ensures public access to administrative record files. This includes notifying the public of the availability of the record, making the record available for public inspection, coordinating with personnel at the facility where the record is located, maintaining an adequate copying facility and maintaining a log of persons reviewing documents. The incumbent will have to respond to phone calls and visitors wanting information on and from the record. These functions will be coordinated with the Office of Public Affairs and Superfund Community Relations Coordinators.
- 7. Maintains the Regional Superfund Central Library of guidance documents and technical references.

CONTROLS OVER WORK

The incumbent works under the general supervision of the [Hazardous Waste Branch Chief]. An administrative record is reviewed and certified for litigation by a person designated by the Regional Administrator.

APPENDIX E

COMPENDIUM OF CERCLA RESPONSE SELECTION GUIDANCE DOCUMENTS

The "Compendium of CERCLA Response Selection Guidance Documents" is a collection of guidance documents which are frequently used in the selection of a CERCLA response action.

The following documents are the core guidance documents used in the CERCLA response selection process and are included in the Compendium. The Agency is preparing a more comprehensive list of frequently used guidance documents. This particular list and file structure was adapted from "Selected Technical Guidance for Superfund Projects," September 1988 - OSWER Directive No. 9200.7-01.

SITE ASSESSMENT

Data Quality Objectives for Remedial Response Activities: Development Process [EPA 540/G87-003]

Data Quality Objectives for Remedial Response Activities: Example Scenario [EPA 540/G87-004]

Design and Development of Hazardous Waste Reactivity Testing Protocol [NTIS PB-84-158-807]

Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA (Interim Final) [OSWER Directive 9335.3-01]

Handbook for Evaluating Remedial Action Technology Plans (Final Report, Oct. 81-Jun. 82) [NTIS PB-84-118-249]

Handbook, Remedial Action at Waste Disposal Sites, Revised [EPA 625/6-85-006]

Modeling Remedial Actions at Uncontrolled Hazardous Waste Sites [OSWER Directive 9355.0-08]

Test Methods for Evaluation of Solid Waste: Physical/Chemical Methods, 3rd Edition [GPO 955-001-00000-1]

MONITORING AND SAMPLING

Geophysical Methods for Locating Abandoned Wells [NTIS PB-84-212-711]

Geophysical Techniques for Sensing Buried Wastes and Migration [NTIS PB-84-198-449]

Practical Guide for Ground-water Sampling [NTIS PB-86-137-304]

RCRA Ground-water Monitoring Technical Enforcement Guidance Document [NTIS PB-87-107-751]

Sediment Sampling Quality Assurance User's Guide [NTIS PB-85-233-542]

Soil Sampling Quality Assurance User's Guide [NTIS PB-84-198-621]

SOURCE CONTROL: ALTERNATIVE TREATMENT TECHNOLOGIES

Directory of Commercial Hazardous Waste Treatment and Recycling Facilities [NTIS PB-86-178-431]

Guidance Document for Cleanup of Surface Tank and Drum Sites [NTIS PB-87-110-672]

Guidance Manual for Research, Development, and Demonstration Permits Under 40 CFR, Section 270.65 [NTIS PB-86229-192]

Handbook for Stabilization/Solidification of Hazardous Waste [EPA 540/2-86/001]

Management of Hazardous Waste Leachate [NTIS PB-81-189-359]

Mobile Treatment Technologies for Superfund Waste [EPA 540/2-86-003F]

Practical Guide - Trial Burns for Hazardous Waste Incinerators: Project Summary [NTIS PB-86-190-246/AS]

Prohibition on the Placement of Bulk Liquid Hazardous Waste in Landfills - Statutory Interpretive Guidance [NTIS PB-86-212-271]

Review of In-Place Treatment Techniques for Contaminated Surface Soils, Volume 1 Technical Evaluation [NTIS PB-85-124-881]

Review of In-Place Treatment Techniques for Contaminated Surface Soils, Volume 2 Background Information for In-Situ Treatment [NTIS PB-85-124-889]

Systems to Accelerate in Situ Stabilization Waste Deposits [EPA 540/2-86-002]

Technology Screening Guide for Treatment of CERCLA Soils and Sludges [EPA 540/2-88/004]

Treatment Technology Briefs: Alternatives to Hazardous Waste Landfills [EPA 600/8-86-017]

SOURCE CONTROL - STORAGE/LAND DISPOSAL

Systems to Accelerate in Situ Stabilization of Waste Deposits [EPA 540/2-86-002]

GROUND-WATER REMEDIATION.

Guidance on Remedial Actions for Contaminated Ground Water at Superfund Sites (Interim Final) [OSWER Directive 9283.1-2

Leachate Plume Management [NTIS PB-86-122-330]

RISK/ENDANGERMENT ASSESSMENT

Superfund Exposure Assessment Manual (Draft) [EPA 540/1-88/100]

Superfund Public Health Evaluation Manual [EPA 540/1-86/060]

CONSTRUCTION TECHNIQUES AND PROCEDURES

Emergency Response Procedures for Control of Hazardous Substance Release [NTIS PB-84-128-719]

Guidance Manual for Minimizing Pollution From Waste Disposal Sites [NTIS PB-286-905]

Guide for Decontaminating Buildings, Structures, and Equipment at Superfund Sites [NTIS PB-85-201-234/AS]

Slurry Trench Construction for Pollution Migration Control [NTIS PB-84-177-831]

Superfund Remedial Design and Remedial Action Guidance [NTIS PB-88-107529]

Superfund Removal Procedure Revision #3 [OSWER Directive 9360-0.03B]

PROGRAM GUIDANCE

CERCLA Compliance With Other Environmental Statutes [OSWER Directive 9234.0-02]

Guidance Document for Cleanup of Surface Impoundment Sites [NTIS PB-87-110-664]

Handbook, Remedial Action at Waste Disposal Sites, Revised [EPA 625/6-85-006]

Superfund Federal-Lead Remedial Project Management Handbook [EPA 540/G-87/001]

Superfund State-Lead Remedial Project Management Handbook [EPA 540/G-87/002]

CASE STUDIES

Case Studies 1-23: Remedial Responses at Hazardous Waste Sites [NTIS PB-86-121-721]

Summary Report: Remedial Response at Hazardous Waste Sites [NTIS PB-86-121-721]

COSTS

Costs of Remedial Response Actions at Uncontrolled Hazardous Waste Sites [NTIS PB-83-164-830]

Remedial Action Costing Procedures Manual [NTIS PB-88-113-691]

Removal Cost Management Manual [OSWER Directive 9360.0-02B]

APPENDIX F

MODEL TRANSMITTAL COVER LETTER

[Name of Contact]
[Address]

Dear [Name of Contact]:

The U.S. Environmental Protection Agency is required by law to establish administrative records "at or near a facility at issue." This administrative record consists of information upon which the Agency bases its selection of response action for a particular Superfund site.

By providing the public with greater access to these records, it is our hope that they will be better equipped to comment constructively on site activities and to understand the issues relating to the selection of the response action at the site.

We appreciate having the [Name of local repository] as the designated administrative record facility for the [Name of site] Superfund site. The enclosed record files, along with any future documents relating to technical activities at the site should be placed in the [Name of local repository] and be available for public review. The record files should be treated as a non-circulating reference - it should not be removed from your facility.

Also enclosed is a fact sheet to assist you and your staff in answering questions posed by the public concerning administrative records for selection of response actions at Superfund sites. Please feel free to distribute this guide to the public.

To ensure the receipt of the administrative record file, I would appreciate your completion of the attached Document Transmittal Acknowledgment form. Please return this form in the enclosed self-addressed, stamped envelope.

Again, I would like to thank you for your cooperation with the U.S. EPA in serving as a Field Repository. If you have any questions or comments, please contact [Name of EPA contact] at [Phone No.].

Sincerely,

[Name]
Administrative Record Coordinator

APPENDIX G

MODEL DOCUMENT TRANSMITTAL ACKNOWLEDGMENT

From:	[Regional Office Address]	
To:	[Field Repository Address]	
the U	nowledge that I have received the following some site.	
•	Administrative Record Name -	[Site Name]
	Administrative Record Document Numbers -	
		1.
	<u>.</u>	
Signe	α	
Date		
Pleas	e return this form to: [Regional Office Ac	idress]

APPENDIX H

FACT SHEET

Administrative Records in Local Repositories

The "administrative record" is the collection of documents which form the basis for the selection of a response action at a Superfund site. Under section 113(k) of the Comprehensive Environmental Response, Compensation, and Liability Act, as amended by the Superfund Amendments and Reauthorization Act (CERCLA), EPA is required to establish an administrative record for every Superfund response action and to make a copy of the administrative record available at or near the site.

The administrative record file must be reasonably available for public review during normal business hours. The record file should be treated as a non-circulating reference document. This will allow the public greater access to the volumes and also minimize the risk of loss or damage. Individuals may photocopy any documents contained in the record file, according to the photocopying procedures at the local repository.

The documents in the administrative record file may become damaged or lost during use. If this occurs, the local repository manager should contact the EPA Regional Office for replacements. Documents may be added to the record file as the site work progresses. Periodically, EPA may send supplemental volumes and indexes directly to the local repository. These supplements should be placed with the initial record file.

The administrative record file will be maintained at the local repository until further notice. Questions regarding the maintenance of the record file should be directed to the EPA Regional Office.

The Agency welcomes comments at any time on documents contained in the administrative record file. Please send any such comments to [name and address]. The Agency may hold formal public comment periods at certain stages of response process. The public is urged to use these formal review periods to submit their comments.

For further information on the administrative record file, contact [name and phone no. of Administrative Record Coordinator].

APPENDIX I

MODEL NOTICE OF PUBLIC AVAILABILITY



THE UNITED STATES ENVIRONMENTAL PROTECTION AGENCY ANNOUNCES THE AVAILABILITY OF THE ADMINISTRATIVE RECORD XYZ SITE, [Locality, State]

The U.S. Environmental Protection Agency (EPA) announces the availability for public review of files comprising the administrative record for the selection of the [remedial, removal] action at the XYZ site, [Locality, State]. EPA seeks to inform the public of the availability of the record file at this repository and to encourage the public to comment on documents as they are placed in the record file.

The administrative record file includes documents which form the basis for the selection of a [remedial, removal] action at this site. Documents now in the record files include [preliminary assessment and site investigation reports, validated sampling data, RI/FS workplan, and the community relations plan]. Other documents will be added to the record files as site work progresses. These additional documents may include, but are not limited to, the RI/FS report, other technical reports, additional validated sampling data, comments and new data submitted by interested persons, and EPA responses to significant comments.

The administrative record file is available for review during normal business hours at:

[Repository Name] and U.S.EPA - Region Z [Address and Phone #]

Additional information is available at the following locations:

Verified sampling data - Contract laboratory,
 and documentation [Address and Phone #]

Guidance documents and - EPA-Region Z technical literature [Address and Phone #]

Written comments on the administrative record should be sent to:

[Name], Office of Public Affairs
 U.S. EPA - Region Z
 [Address and Phone =]

APPENDIX J

MICROFORM APPROVAL MEMORANDUM



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

OCT 2 | 1988

OFFICE OF
SOLID WASTE AND EMERGENCY RESPONS

MEMORANDUM

SUBJECT:

Microfilming the Administrative Record

FROM:

Edward J. Hanley, Director

Office of Information Resources Management

TO:

Asa R. Frost, Jr., Director

OSWER Information Management Staff

In accordance with EPA Records Management Manual, Chapter 6, dated 7/13/84, I approve OSWER's request for an administrative record micrographic system for regional hazardous waste management programs.

The feasibility study prepared for OWPE, entitled "Assessment of the Suitability and Costs of Alternatives for the Administrative Record" (June 30, 1988), satisfactorily documents and justifies the need for converting the administrative record to microform. In particular, the requirement under SARA to make the administrative record publicly available at or near each hazardous waste site makes microform a cost-effective storage medium.

Please inform each regional hazardous waste program of my approval of OSWER's request and of the need to comply with the remaining provisions of Chapter 6 of the EPA Records Manual should the region proceed with implementing an administrative record micrographic system.

cc: SIRMOs, Region 1 - X

APPENDIX K

MODEL CERTIFICATION

IN THE [NAME OF COURT]

UNITED STATES OF AMERICA,

Plaintiff,

v. :

[NAMES OF DEFENDANTS]

Defendants, : CIVIL ACTION

NO. [number]

v.

[NAMES OF THIRD PARTY
DEFENDANTS]

Third Party Defendants

CERTIFICATION OF DOCUMENTS COMPRISING THE ADMINISTRATIVE RECORD

The United States Environmental Protection Agency (EPA) hereby certifies that the attached documents constitute the administrative record for selection of response actions under the Comprehensive Environmental Response Compensation and Liability Act or 1980, as amended, for the [name of site] site in [City or County], [State].

By the United States Environmental Protection Agency:

subscribed my	y name	this		
	signati	urel	· .	
[1	typed	name_		

act in their espacity of a writ of government, they may receive funds from the fund for estade 10% response action to authorized by eaction 111(a)(1) of CERCLA. A political authorizing of a State in treated of a State for the purpose of section 107.

3. Grants for tochnical assistance.
Current NCP § 200.29(d) rolors to
cooperative agreements and contracts.
Amendments to CERCLA section 111
authorize technical assistance grants
pursuant to section 117(e). Cooperative
agreements and grants, when taken
together, are generally referred to as
"assistance agreements." EPA is
proposing to revise § 300.23(d) to refer to
"procurement contracts or assistance
agreements."

Subpart I—Administrative Record for Selection of Response Action

Proposed Subpart I of the NCP is entirely new. It implements CERCLA requirements concerning the establishment of an administrative record. Section 113(k)(1) of CERCLA requires the establishment of an administrative record that contains the documents that form the basis for the selection of a CERCLA response action. In addition, section 113(k)(2) requires the promulgation of regulations establishing procedures for the participation of interested persons in the development of the administrative record.

EPA is proposing regulations regarding the administrative record that include procedures for public participation. This will ensure the development of a complete and accurate record by all parties responsible for compiling records, because procedures for establishing and maintaining the record are closely related to the procedures governing public participation.

Because this subpart is entirely new. the following discussion is not divided into major revisions, other revisions, and points of clarification. Instead, it explains the purpose of the administrative record and then generally provides a paragraph by paragraph explanation of the proposed regulations.

A. Bockground and Purpose

Under CERCIA, the administrative record cotablished under section 113(k) serves two primary purposes. First, under section 113(j), judicial review of any issue concerning the adequacy of a response action is limited to the administrative record. Second, section 115(k) requires that the administrative record be used as a vehicle for rublic participation in the selection. A has

considered all relevant factors in coloring the response and that interested parties have been given adoquate notice and an opportunity to participate in that colorion.

1. Judicial review. Section 113(j)(1) of CERCLA provides that judicial roview of Day isouce concerning the edequery of any rosponer action shall be limited to the odminiotrotive record. Section 113(j)(2) provides that the court shall uphold the eclection of a response action unless the objecting party can demonstrate, based on the administrative record, that the decision was arbitrary and capricious. or otherwise not in accordance with law. These sututory provisions codify wellcotablished principles of administrative law concerning the applicable mandard and acope of review for informal allowed actions. The legislative history of section 113 demonstrates that it is intended to clarify and confirm the applicability of those administrative law principles to CERCLA response oelection. (Seo S. Rop. 82–11. 89th Cong.. 1st Sess. 57 (1985); K.R. Rop. 82–233, 89th Cong., 101 Seco. 82 (1998): Cong. Rec. H 11084 (daily ed. Dec. S. 1885)).

Limiting judicial review of the selection of a response action to the administrative record ensures that litigation on the colection of the response action focuses on the selection in light of the information available to the decisionmaker at the time the response was selected. Judicial review limited to the administrative record contributes to the overwhelming public interest in effecting the expeditious cleanup of potentially health- and environment-threatening hazardous waste sites and ensures that all interested persons may participate equally in the administrative decisionmaking process. The principal effect of limiting judicial review to the odministrative record is that cours will not engage in de novo fact-finding during their review of a challenge to the decision to select a cortain response. Thus, record review of response relection decisions would mean that persons challenging the response decision could not depose. exemine or cross-examino on-seens coordinaton (OSCs), remedial project managem (RPMs). government compulsate, or decision malkers with respect to the response decision or engage in any other discovery activities. Also, the imposition of long and costly trial-type procedures in section 103 actions would greatly

2. Public participation. Sections 113(k)(2) (A) and (B) of CERCLA require the promulgation of regulations establishing procedures for the participation of interacted percent in the development of the ediministrative resert. Participation by interacted person, when appropriate, will caute that EPA has considered the concerns of the public including percentally represents pertice (FRPs), in extecting the response oction. In addition, for purpose of ediministrative and judicial review. The ediministrative record can contain decreases that reflect the views of the public, including PRPs and those not party to any judicial proceeding. concerning the coloction of a response oction.

For remodial actions, eaction 119(h)(2)(B) of CERCLA cotablishes the following minimum precedent for public nurscipation:

i. Notice to percaptally affected persons and the public, accompanied by a brief analysis of the plan and alternative plans that were considered:

ii. A reasonable opportunity to comment and provide information regarding the plan:

iii. An opportunity for a public meeting in the affected area, in accordance with exciton 117(1)(2) of CERCLA:

ty. A response to each of the cignificant comments, entrings, and new data submitted in written or oral presentations: and

v. A statomont of the back and purpose of the calected action.

These requirements are virtually the came as these required by coction 117 of CERCLA concerning public participation for remoded actions. These public participation requirements are proposed for codification today in § 200.439 of Subpart E of the NCP. Subpart I ampands on the public participation requirements of Subpart E.

වියදෙනය එම කරණත මේ හැකුමෙන් රදස්තය often involves the read for prompt action, the procedures proposed teday for public participation in removal ටෙන්නෙන් කණේ ක්කේට්ඩ්වේ රාක්ෂ වානයේ වෙන අත්ත්ර romodial actions. Ramoval authority വിയോ ശാ 1004 വ്യാമത്യ ശ ഇറോ ഉയില്പു ta situstions where prompt load agency Detion is warrance Section 113(L)(2)(A) of CERCLA requires that there be "appropriate" participation of intersolet ഉത്തരമാ ഇ കോ പ്രാത്തുവേട്ട ഉദ് വരം Compared in page 23 CARCER (Marie Page 1 rozevol actions. The legislative biotory of this section orates that these public porticipotion requiremento "oro not intended to homper emergency romoval actions. Nonetheless, the Administrator in directed to develop appropriate puricipation procedures for removal actions and should follow these Ledith Center of the wormany cuisat practicable." (M.R. Rep. 63-233, 1314)

delay response.

Cons. 1st Som 1972 at ESL Public participation regression for regression actions are commend in [200.415(a) of oday's proposed regulations. Additional cublic porespines precedures is the development ed co commingo avo record for a removal assisse are addressed in 9 300.520. The public perticipation no studens of bangicole see seasons on appropriate byd of public involvement for removal accoust without cousing innecessity delay. In general, where there in time to colicit public comment refore the existion of a removal action. the lead agency will do so. Public participation procedured for reserval actions on described in proton describ celow.

3. Current Record Requiremento

Section 113(k)(2)(C) of CERCLA states that until regulations on the participation of interested persons in the ievelopment of the administrative ecord are promulgated. the administrative record shall coasist of all temo developed and received pursuant o current procedures for selection of the esponse action, including procedures or the participation of interested parties and the public. Current procedures on subjection in the selection of esponoe ochens include en extensive danced accounties diaments which interested persons have monce of nformation through notices in local newspapers. community relations nailings, public meetings, and letters. ncluding notice letters to potentially esponsible sarces. An adequate record nould be compiled and maintained brough use of surrent procedures for es where the remedial investigation or emoval action bas already begun pro .: o promulgation of these regulations. These proposed administrative record equirements build upon and formalize existing procedures for the exchange of mormation on the relection of a esponse action.

The cutoff date for the applicability of hese regulations is based as whose the diministrative record file must first be node evaluable under these regulations. The lead agency may not be able to fully amply with regulations concerning amplication of the record which are romulgated after a record has already seen compiled and made available at or lear a site. Thus, at such sites, the lead gency will comply with these explotions to the entent practicable.

. Summary of New Subpart I

1. Establishment of an administrative ecord (§ 200.600). As explained earlier, ection 113(k) requires the establishment for administrative record consisting of its documents that form the basis for

the pelection of a response action. An administrative record is the compilation of documents considered or relied on by the agency in making a decision: in this case, the selection of the response actor for the cite. Proposed (200.600(a) codifice this statutory provision and provides that such establishment is the responsibility of the lead agency. The regulation also uses the term 'administrative record file" to r : - r to documents which the lead agency anucipates will be included in the administrative record when the decision on response action selection is made. The administrative record file contains a body න් documents කත්ස් incresses යා documents are added and does not necessarily constitute the final edminismive record.

The term "documents," also used in the preamble and proposed regulations, is intended to be very broad. It includes writings, drawings, graphs, charts, photographs, and data compilations from which information can be obtained. It does not include physical camples.

Section 305.800(b) addresses administrative records for Federal facilities, Executive Order 12580 authorizes Federal agencies to establish the administrative record for pelection of response actions for Federal facilities under their jurisdiction, custody, or control. EPA. bowever, is required to promulgate regulations establishing procedures for the participation of interested parties in the development of the record. Federal agencies must compile and maintain records as required by this subport as finally romulgated. Section 300.800(b) also arifies that although the Federal agency is responsible for compiling and mountaining the administrative record. EPA may furnish documents which the Federal agency is to place in the administrative record file to easure that the administrative record includes all documents which form the basis for the oclection of the response action.

Section 300.8CO(b)(2) provides that when EPA (or the United States Const Guard (USCG)) is the lead agency at a federal facility. EPA (or USCG) aball compile and maintain the record.

Executive Order 12560 delineates exact for USCG) in the lead agency. EPA is the lead agency. EPA is the lead agency. For example, at federal facilities conducting on-oite emergency removal actions (other than at DOD or DOE facilities). The USCG can be the lead agency at ... Federal facilities with on-oite emergency removal actions in the constal some.

Section 300.000(b)(3) requires that when EPA is involved in the selection of a response action at a Federal facility on

the NPL the Foderi areas vivil provide EPA with a copy of the index of documents included in the administrative record file, the RI/F3 workplan, the RI/FS released for public comment the proposed plan, any public comments received on the RI/FS and propossed plan. and any other documents requoned by EPA on a case-by-case books. EPA is involved in the selection of o respense oction when it is jointly ociceway with the Federal ogazacy. Oo deliacoted ia Exceutivo Order 12800. Such joint ocicetian ocean. for example, for all remedial actions at Federal facilities on the NPL la ouch coocs. EPA must be cufficiently familiar with the contents of the edministrative record to be able to ocient joinely the response cetion.

PA comidered other options for involvement in the development of the administrative record for Federal facilities, such as penodic visits to the Federal facility to review the administrative record file as it is compiled receipt of the entire contemp of the meard file for all NPL sites. and receipt of the entire contents of the record the for all response actions at all Federal Leilities. EPA bas tentatively rejected those option as being overly burdensome. EPA becieves that the haves cwells notice bearing flexibility for EPA to easure that the response petion selected by the Federal agency adequately accounts for the concerns of the public is consistent with response action seiection at non-Federal facilities and allows EPA to bo oulliciontly involved in the decision when it is jointly solocus, the response ocues. EPA coliciu commento or alternative procedures for EPA's involvement is the development of the administrative record for Federal

Section 200.600(c) openities that it is the responsibility of the State to compile and maintain administrative records at a Stato-load site. Soction 200.200(c) applies only if EPA and the State formally designate the State as the lead agency for a cito as specified in Subpart A wader the definition of lead agency. The requirements for State-load sites are pizallar to those for Federal agencies compiling administrative records for Federal facilities at which EPA is convoived in the coloration of the response action. EPA is proposing that the State provide EPA. compacacing at the time the administrative record file is first made available to the public with the index of documento included in the administrative record file. The incues relating to this requirement are similar to those outlined above for Federal

facility. Additionally, DA may require that States place additional decumants in the record file to come that the administrative record includes all documents which form the basis for the selection of the response action.

Section 300.809(d) provides that Subpart I applies to all response actions takes under section 10% of CERCLA or sought occurred or ordered Odministratively or judicially under section 100 of CERCLA. The ototutory language of section 113(j)(1) states that in any judicial action under this Act. judicial review of any issues concerning the adequacy of any response action taken or ordered by the President shall be limited to the administrative record. It has been argued that section 113(j)[1] of CERCLA does not apply to injunctive actions taken under section 196, and that the literal meaning of the phrase "taken or ordered by the President" does not include section 10d actions for injunctive relief unioss on administrative order is issued.

The statetory language of sections 113(j) (1) and (2), when read together, indicates that this narrow interpretation of sections 113(j) (1) and (2) provide that judicial review of any response action is limited to the administrative resord. In addition, section 121 of CERCIA expressly provides that the Provident shall select all remedial actions to be carried out by EPA under section 103. No exception for section 103 injunctive actions was made.

Accordingly, consistent with the statutory large-upge and congressional intent. EPA is clorifying that limiting judicial review of response action selection to the ediministrative record applies to all actions taken under section ICA of CERCIA or cought secured, or ordered administratively or judicially under section ICA of CERCIA.

Section 300.800(d) further provider that Subpart I saly applies to those often at which the remodeled beveregotion **ോയുള്ള പാട്ടാ ടെ സ്റ്റ** റല്ലേള്ള <u>യായുന്നുള്ള</u> in signed after the passed attents of them regulations. For these sites grandiothered by panapph (d). paragraph (c) provides that the lead agency chall comply with thece regulations to the extent proceeds on a case-by-case basis. This deep nec sac appears systematical and according to the අරුත් දුල රැවැට පැවැති අත්ව ම්පැවැත අත judicial review of the externion of a response action at these often will not be limited to the administrative record Rather, as explained earlier, this provision simply recognizes that there will be ongoing actions at water the final regulations causet be complied

with in full. The public puridepusses procedume for reserving location sullimed in which is reserved and 117 of the state of the course of continue in the present of the course of the

Subpart I does not apply to third party clanguas, i.e. those not waderaden pursuant to socions 124 100 or 111 of CERCLA Under this proposal such cleanups need not comply with these oquiantoria ecolo mantebero. Section 300,000(d) එකෙ සහ හලාණ ස්ථාව State according to cost recovery words cocios 107 of CERCLA whose the Sule and only in own chipocinco to covering o response some and with this oubpart If a State in english to encor... costo from responsible person under section 107 of CERCLA. EFA may with to require that I were comply with this oubpart to curredite judicial proceediage in such circumstances. EPA solicies coනയമാന മേ കുടന്റെലു സ്വാധ ധിപ്പാദ്യമാ espectation product of views blunds.

2 Location of the administrative record (§ 300.003). Section 113(t)(1] of CERCIA requires that "the odana sectivo record skall bo available to the public of or moon the facility of ione. The President also way place duplicates of the administrative record of any other location." EPA proposes to require that the administrative rocord file generally be located to two places. First. as provided by the stender. the record file obali be leconed or or now the locality of issues. To estadorna to the terminology of the rest of the NOP. the भागनीय एक्टी खं किएक एवं प्रांत "Otio" ब्याभा ලා දු ග්රේක්ෂය දින් දෙන් දෙන් ක්රීම්න ද යෙ (නණ්තරෙන්ව ක් මනසා

la oddisa. EFA proposa libe like administrative කදහල් කි්දු හා මහෙලේ යැ ටත රුහුරෙ න් අතුරු අවල් ට්ඨාන්දුර් නේ දෙස්යෙ දෙකුනා | ලොන්නෙ විගුලකුම් රා රේ දෙකුනා | loco ත්තෝ කළුප්තර ගත මුල්ක සියල්කෙක් Office. ගෘ මී?A කිටල් ඉකිහ. c Fශ්යෝ ogracy resivolent to on EPA Regional office. or. for State local critica o Slate cavironaacabi පැපැතෙ ශ්රීය. ම්ªA concidental ചവച്ചതു ക്രാ ത്ഷേബി lത്കേർത്ത ted between tre decides measuring a අතර ලක්කම පාලක මෙන මෙනෙක් ලක්ක ATO ENGINEER OF AND SECTION AND SECTION OF A **രാഗങ്ങായെ വ ദ്രാ ഗോളപ്പെ പ്രാവുവ** മുറത at or near the cite. As decembed below. ණීට මුල යැය නැය ඇට ශ්ය දැනමුල් මෙන්නුන ු දෙනු න් කනැ ඒ ණා එකෙනෙන් included in the education with the included file at the execute lecension.

Under (1 200.005, the file of the exact location must contain all documents which are part of the administrative record except extens without the files date. Quality contains and quality

chompost geometric entropy polect.

മാ 10 വി ക്രോഹ റത്തെവ്ത്തൽ വേറ് മാവം നൂറ വുന്ന വാവു സായിയുട്ട ഉവ 01 അറ ത് ഇറ അത്രം അത്രാവിക്കുറാ യുട്ടി ജവു വിഹാര്യ രാൻ ഉപ്പെട്ടിയുട്ട with the rely accorded another responitery, acceptanced by the **community** നിഥത്തെ ഇടഞ്ഞവാം. ഇറ്റ centria additional information which in **അത്** റ ഉത്ത് (ജ എറ ഹാള്ളോ റ വഗ്രെ ഇത്തെ നേള്ളുന്നു പ്രവാദ പ്രദേഗാവ അർ ഇത്രക്കാരുത്ത് അവയായി അവ MPL libers is the said in the control community പ്രശ്യം പ്രത്യേഷ്ട്ര നുമാനു വാധി തിരുത്തിലും വുദ്ധിക്ക റ്റ് ഉത്ത്യേരുക്കുട്ടി വ്യാത്രാപ്പെ വ රුණාන්වන කරන කරන්න වැන්න කරන මුල්ල කරා අ be located in come other publicly accombinates PA in seemed adaptives റട**്ടെൻ ഉള്ളാക്കാരി ഈ അട്ടാട**്ടാ ३० त्रवित्यीवराठ दर्व इंडम्ब्ल विशंबंत्र ब्रब्बेट ब्लब्बेखां രെ മറവു ഇവാ ഇ അവരുക്കു വലവു ക്ര प्रहार के उन्हें प्रकार कि उन्हें के उन उन्हें के उन्हें ace who here we say the heart librusia. Lenz bulla es probles colores ගැන නැලේ යා මනුමුල්ව යනෙයනුගැ local some

EPA sony solve the desiration for solvest the local public is suitable to the point of the solvest to the point of the suitable to the point of the suitable to the suitable to

නුවෙන කෙන්න් නෙන අර්ථ හා නෙනුනේ ඇම් නගැන් නෑ මට ඉවතුන්වෝතු මනොගන් යෑ පෙ മാവം ഈ വള്ള ഉപ്പെട്ടാ വുന്നു പ്രവാമ <u>റർജ്യങ്ങാണൂറ</u>റ ഉക്കുകള് ഉപ്പാദം The identification and continues in a continue of the continues of the con كا تترسك في تصميح فاستها في soldier to silder cel es aldalica (Openios for among () Energy () provides that contain appeal of tochaient حظ ها أعلاما بع ربي وهامهمولها വ നട്ടുപ്പ പ്രത്യാത്ര അതിലേച്ച പ്രട്രമാ ලක්කම ල්ලාවේක් ද ල්ල්ල් මෙන වුණුර राष्ट्रि क्रिकेट प्राप्त स्वर्णान्यां प्रति का उपक्रम යනු නේ ලාදනුවෙන් යේ බැය අවතුය മായ മാവ അർ വിമി ഉപ്പെ വാമത്തിന്റെ අතර ඇර ලකු දැණි ලකුණා! ලකුණා. මකර ල් කණ්ටකෙන්න් රුණු යනස්ස රූ**ුවාන**් located යන් වන හැම වැන මුව අතුත්වයේ දියා ලෙන්ට නොවන් ල්නෙට යන්ගි . පෙන්න දෙන් continues to be costly occasible to in:erested ඉදනකො. ව්යාකමයා ශ් ශස්ත ಪ್ರಾಂಗ್ಯಾಂಚಿತ್ರ ಚಿತ್ರಚಿತ್ರ ಕರ್ಮಕ್ಷಿಣಿಕ ಉಗ್ಯಾಹಿನ್ಡ ർചവ. സ്ക്ഷ് വര മാരംപി y സമ്മൈവര്ക്കാർ ശ

data opposery observed and are quite voluntations. documentation of quality essurance and quality control which is normally opposed in the remodual investigation/feasibility study (RI/FS). and chais of evotody forms. These types of documents may be stored in the EPA Regional office, control laboratory office that conducted the testing. State environmental agency office, or eisewhere, as appropriate.

Section 200.605(b) provides that midence documents not generated for the porticular site for which an cominiowative record to being compiled ent to verteil a at bearinging of vert central location. The guidance documents need not be in each sitespecific administrative record file at the central location or at or near the site at move. EPA anticipates that each EPA Regional office will maintain a central library of maidance documents which are frequently cited as a basis for selecting o response oction. This opproach climinates the need for reproducing copies of the same cocument for each site record. The term guidones document includes issuespecific policy memoranda as well as formal guidanco documento. Examples of ouch midance documents and issuespecific memoranda include the RI/FS guidance document, guidance on mak/ exposure assessments, guidance on applicable or relevant and appropriate requirements. Elemortado on moximum contaminant levels, and guidance on testing for opecific contaminants.

Cuidance documento and memoranda which are generated for a particular site must be placed in the site-opecific administrative record file. (For example a document on district contamination of XYZ site must be placed in the XYZ site-specific administrative record file. If it is also used as a guidance document on the cleanup of district at the cleanup of district in the cleanup of district in the cleanup of district in the control it may be located only in the control library rother than physically in the control will be available to the public.

EPA proposes in § 200.895(c) that publicly available technical literature not generated for a cite at issue aced not be located at or near the cite at issue, in the central library of guidance documents or in the site-specific administrative record file, provided that it is listed in the index to the administrative record. Copyright laws may but the copying of these materials without specific approvals. EPA believes that expending Superfund resources on obtaining copies of publicly available technical literature is

not appropriate. Examples of publicity available technical literature include widely used engineering handbooks on ground-water monitoring, and articles from technical journals, which are readily available in technical libraries. The index must list these documents apparately and indicate information on their availability, or, the literature may already be cited in a document included in the record.

Tochnical literature. however, which is not generally available should be included in the oite-opeciñe administrative record file. Bacause these document on by definition act obsily obtainable, they should not simply be indexed. They generally will not be used for many sites: therefore, it is also not ad an impropriate to include them in the control library of guidance documents. The are should be reserved for documents which are bequently used to select response actions. Examples of technical literature not generally ovollable include orticles from technical journals or unpublished documents act available through the Library of Congress or not circulated to technical libranes.

Section 300.605(d) provides that documents included in the confidential portion of the administrative record file shall be located only in the central location. Since the public cannot review the confidential and privileged information, there is no reason to require that such information be maintained at or near the cite.

EPA in proposing in { 200.205(b)(S) that for reasons of administrative feasibility. an administrative record file for emergency removal actions where on-cite activities cause within 30 days of inivation need only be available for public impection of the central lecation. Emergencies are those actions with little or as look time and generally of very obor durotion—for example, a highway opill. The benefite of picking the record file at or near the cite are ourweighed by the administrative burden on the response to ouch exergencies. Where fensible. a motice way be placed at the svitestaining that the edinioleus stie record file will be available for public inspection at the EPA Regional office (or cther central location).

3. Contents of the administrative record (§ 300.810). The administrative record under section 113(h) consists of documents which form the bosis for the selection of a response action of a particular site. In determining which documents form the bosis for the response action. i.e., what constitutes a complete record, the lead agency shall include all documents considered by the

decisionmaker, including those relied upon by the decisionmaker in selecting the response action.

It should be noted that documents constituting the administrative record for aclesion of a response action are only a subset of documents that the land agency way have compiled with respect to a particular site. The land agency will also have general files consisting of documents relevant to other aspects of a site.

Soction 300.810 discusses generally what abould be corrained in the odminiowody o rocerd filo for rosponse colorion and what should be excluded. Socies 200.610(a) states that it should ලෙකාටක රිජොව ක්රිකෙවයක. එවැර: analysis of the factual information and data: reidanco documents: tochnical literature: cite-opecific policy memorindo: documento received. publicated or mode available to the public under [[] 300.815 and 300.820 of this subpart decision documents: and caiorcement orders. La oddition. On index listing the documents contained in the odministrative record file should be included at the beganing of the record

The following is a list of documents which apically, but not in all cases. should be par of the administrative record for colection of a remedial or removal action. (For purposes of this cubpart en RI/FS chould be included as a component of a remedial action record.) Only documents within each category which form a basia for celecting the response action will be par of the record (i.e. although correspondence is listed under public participodion, correspondence on liability iccues is not part of the record). This list is intended to be illustrative. but not necessarily required at each site വാളിത്തോ പ

t. Contonts of Remedial Action Administrative Record.

(a) Foctual Information/Data.

Sompling plan.

Validated compling and analysis data. Chain of evotody forms.

Project plan or program plan (QAPP).

Proliminary addressment report.

Silc involvention report.

inspection reports.

FI/FS final workplum.

Associates to facil RI/FS workplos.

Summary of romedial action alternatives (used in conjunction with

carly opacial notice lettern). Cata ownmary cheets.

RI/FS.

Tochwich onedien.

Factual information submitted by the public, including PRPs.

Documents supporting the bod offench, o gordumino pies of issuincer org යන්නේ මෙන්නේ විස්ත්රාවේග

(b) Policy and Cuidance.

Mozamaco on policy decisions (sitepecific and bouc-opecific).

Guidance documents

Technical literature.

(c) Public Participation.

Concoppondence.

Public zoreco.

Public comments.

Community relations plan.

Notice letters to PRPs.

Proposed alan.

Transcript of meeting on RI/FS and proposed plan. and waivers under section 121(d) of CERCIA.

Documentation of other public

Response to significant comments. (d) Other Perry Information.

ATSOR Models occomment

Notural Recourse Trustees finding of

fact and final reports.

L'ocusaoniation of State involvement. (e) Decision Documents.

Rossed of Doctron, including

reoperniverseo oursery.

(f) Enforcements Documents.

Adamenta by miss.

Convent degrees.

Alladova

Responde to police iction containing relevant factual istinggange.

(g) Index.

ii. Contenu of Removel Action Administrative Record.

(a) Factual Information / Date.

Sampling plan.

Validated compliag and analysis data

Chain of currody forms.

Proliminary മഗ്ദാവതരൽ report.

Site investigation report.

Inspection roperty. Engracemny evolutions/Cost analysis

resort (EE/CA).

Technical otucios porfermed for the

Factual information submitted by the public incinding PRP0.

Docameu ව වෙම්මන්න්න් දීමට 100 අ ogency o do comination of imminest cad substantial cadoacomeoat

(b) Palicy and Guidener.

Memoranda en pelley decisiona (citaspecific and issuc-opositic).

Canquees docares and

Technical literatura.

(c) Public Perticipation.

Correspondence.

Public novices

Public commonsu.

Community relations plan.

Notice letters to PRPs.

Documentation of other meetings.

Response to significant comments.

(d) Other Party Information.

. Tali ... ausessment

Notwed Resource Truston Ending of lact used back exports.

Documention of State involvement

(e) Docision Documents.

EE/CA opproved memorendum.

Action memorondum. (f) Enforcement Documents.

Administrative orders.

Consent decrees.

Allandaria.

Response to notice letters containing relevent factual information.

(ड्र) ।तर्वकाः

Several decuments in the list above require further explonation. First. verified sompling date one included on the list above. Data which have undergoed quality commonco/quality දෙන නැල් ලක් කරුණුවේ ලක කනාග දිර iකුස්සමයල් <u>iක ශ්ර හදහැන්.</u> ම්රූප සත්සේක තිරුල been rejected as increasing or will otherms as by considered or which ටන් න් ර්ටන්න්න් රජ එමෙ මරටක අතඅග record.

Second EPA is proposing in (300.810(a)(1) that decreases to വേഴുമെയ്ക്കു കോർവരുക്കാര് ഒരു വര ുചാരായ വളർ വളർവാർപ്പെ വേർവുള്ളത്താരു be part of the administrative research EPA and ower Federal agrees wave ൾ വ്യാപ്പുരാ ശ അർട് വാവാം അവ determine കാരുത്തി എന്നു പ്രത്യാത്ത് അർ subcreasiol cardangement to the public posito as suppres as po carmos ses ශ්නය ලෙස අදහන් වේ අත්යය ලෙසක් නේ අතර මේ o berrdow aviouvec. I EPA expers to exercise its discretion to conduct meh an assessment, the occomment civall be included in the moore A decommensuion of an imminent and substantial tousch se pecce or recently information which forms a body for the selection of the response estion. As such, when a datesmination of an imminent and cuberasual endangermant io කාල්ය. it is ඉලුලු මේ කිල හෙදාල් මේ කිල ASS assess caroques o lo coussise believes that judicial serious of the decemiention that than is an imminant and substantial cadangoment in actions under section 103 to entores an order or for injunctive which therefore, is limited to the administrative record.

Third, for a reserved action reserved the list includes a summary of remodial ocuos dierengivos This cussory will only be generous in conjunction with or ceial notice lettern EPA may insue to PRPs pursuant to common 122(c) of CERCLA if the action buter is isoured prior to the availability of an AUFS report and it appears accounty to inform intercoted persons of the lead agency's direction on remedial alternatives. In this context, a summary of remedial action alternatives would be generated if necessary to enable TRPs to make an informed good faith offer to undertake ine remeulal design ur

macdial action. The manner of അവ്വ വായ്യാ വിശ്ചായ്യാ വിശ്ചാ brosom ovincentations and at behalled රු ගෝ ගෙන් වෙන වෙන්නේ රුණු හරන් පො ෆ්හී PRPs have the information.

Fielly, EPA is proposing that socie letters to PRPs be included in the rbaces and AEB breen ovitor winder නොන් ලාශ්ග රජ ලා පතෙව්ග විභාගය iscured under section 122(c) of CERCLA 53 FR 5323 (Fobrany 22, 1823). PRP thor receive police lorem are expected to bocome framing that CERCLA if they have set dready dose to la light of socies brown and gazaral principlos of administrativo law. PRPs are of notice that an eximinismous exected file will ba or in available for public inopaction.

දියන්න දෙන (ල්)මල නම්නානා ed see that Albacett apides assessed included in the ediminimative accord. വ് ഉദ്ദേശങ്ങൾ വരാത്താരു പ്രദേശങ്ങൾ വര്ഷ്ട്ര ye desider cross are (d Decrees) രു വസത്രത്തിന വരാ വരാ അഭ്യമുത്തിനെ inclusion in the edministrative record න් ලෝ ලෝ ලෝ ලේ මෙන් රෙන වේ අවරුව ලෝ මෙන receive of the response casion. There ල් ලෝ ල්ට වියල අප අප අප අප අප multion for purpose of during.

Draft decreesely toward cresonade. and go i-to-go h source of orage Consulty. حل والمخولات والمع الأس Across Served Street of don't उसी हा है छेटा प्रसार है जो उस्कार है जिस है अपने उसका है जिस है අපල පදන්න නො නැකෙනු පත්තරන්ගන්ගන්ගන්ගන්ග season ai eo bailm so basbicaes onsw oction calection and nover reportation by a final decrease and boar base දෙනාවාන කහැපැවැ විදහා පැහැකි එම කගැ pepulaci recensos rodro ken si soodeo in the odminiosective record file. The general rule, bewever, is that ealy final cal si bobulssi cd Les cansausob odministrati - resert

වියකුත්යා :ා වෙනුත් කැකුලෙන්න ගන් day-to-day botes of stall which ear bot مورع مع معرست عود المناموم بع والم റർമ്മുമുന്നു സാരൂർ വന ഉടെയ്യമാരുന හා යනුමාන්නුම හැරුන්නුය හැ ഭരമേയത്യായുടുത്താ ത്യ വസ്ത്ര് ശ രൂശവ വലച്ച or monograph or interpolation വസ്യവം വടന്നു ആവാരു പ്രവേശവം വ aliamativa viampointa

Section 200230(c) addroom privileged decuments Europice of privileged decuments include, but are not limited to: documents or believed to oliomay-clical pavilons and ollomoy More preduct oxelucion excusors orbicet to deliberative precess privilege. വമർ റമ്മിത്രോത്രാൻ ഗാമാൻസ് വിമത്തവർത്ത് Commen law and other privileges being ps accorded

An assertion of confidentiality of information does not necessari! eliminate the poor of promise ouch information part of the administrative recura, il comidonua, inidemation weren

forms a busis for the celection of a responde detion is not included in any orner cocument in the administrative record. that information must be part of the administrative record. Section 300.810(d) requires that the information. to the extent loosible, must be ാധനമാവാർ ia ouch വ അവമരുടെ വാ to make it disclossible to the public and placed in the edministrative record file. If it is not foosible to our marrize the information in a releasable manner. e.g.. when the privilege applies directly to the information which forms a basis for the selection of the response action, such as confidential business information, the documents must be maintained by the lead agency in a confidential portion of the administrative record file. (These documents may be reviewed in comero in any subsequent judicial proceeding.) The index to the administrative record must list the confidential or privileged cocument even though the document will not be available for public inspoction. Whether or not the information can be summarized in a releasable manner, the actual document containing confidential or privileged material must be included in the confidential portion of the administrative record file. In light of the nature of the information in the RI/FS and underlying documents and the fact that contomination levels are generally not privileged. this is not expected to cccur frequently.

It should be noted that section 104(e)(7) of CERCLA governs the extent to which information may be claimed confidential by persons required to provide information under that section. Where confidential business information is claimed. EPA will proceed according to regulations set forth in 40 CFR Part 2.

4. Administrative record for a remedial action (§ 300.815). Section 300.815(a) provides that the documents included in the administrative record file for a remedial action shall be available for public inspection of the commencement of the remedial investigation phase. Generally, the commencement of the remedial investigation phase occurs when the final RI/FS work plan is available. The regulations do not specify when the remedial investigation phase commences because this may be a sitespecific determination. EPA solicits comments on whether the regulation should opecify in greater detail when the lead agency must make the administrative record file for a remedial action available for public inspection. The file at that time should contain the documents which will form a basis for

the pelection of the response action generated or received through the date when the administrative record file is first made available. Documents generally avails: .e when the RI/FS work plen is a: _ oved include a preliminary assessment report site inspection report the RI/FS work plan. underlying inspection reports, and the community rolations plan. From that time until the ROD is signed (except as provided in { 300.825. described below} documents which form the bosic for the celection of the remedial oction, aball be added as generated or received to the administrative record file.

The lead agency may cotablish a system allowing for periodic review of document where there are questions as to whether the documents must be included in the administrative record file. Quarterly or monthly updates of the administrative record file may be appropriate in given situations and allows the lead agency to analyze data and organize it in a manager that will be meaningful to the public. In addition, it may cove the lead agency the time involved in making daily or weekly sidencites of satisfier of cultimentals document chould be added to the administrative record file. If there is no question that a document belongs in the administrative record file. e.g., the RI/FS report the document should be placed in the record file as soon as practicable after its generation or receipt.

EPA proposes in { 300.815(a) that the lead agency publish a notice of availability of the administrative record file. The notice must be published in a major local newspaper of general circulation. as is required for the notice of availability of the proposed plan. (See § 300.430 of today's proposed rule.) EPA considered proposing that a notice be published in the Federal Register for wider circulation, but rejected auch a requirement as unnecessary. EPA coliciu commento on whother a notice of availability of the record or of commencement of the public comment period should be published in the Federal Register. EPA also considered proposing that a separate mothication of known potentially responsible parties be made. Section 113(h)(2)(D) of CERCLA provides that the Prosident chall make reasonable extern to identify and notify PRPs as early as possible before celection of a responce action-EPA will be issuing nouce letters to PRPo under section 122(e) of CERCLA early in the process in many situations. Given these early efforts, as well as the notice in a local newspaper. EPA chose not to propose a separate notification of PRPo here.

Section 370.315(b) clarifies that interested perions may submit comments for inclusion in the ediministrative recerd file during the public comment period on the RI/FS and proposed plan described in § 300.430(f) of Subport E. The load agency need not however, respond to esaments that were submitted prior to the public comment period on the proposed plan although in many instances, the lead agency will cither make appropriate medifications to the response action or respond in writing to those early comments.

A writes response to dignificant comments will be included in the edministrative record file. The lead offercy need not respond to any comments resided entire the coor of the public comment period. Generally, responses will be included in the responses will be included in the responsiveness summers, which is part of the ROD. In responding to dignificant comments, the lead agency need not respond coparately to each comment but may combine comments by oubject or other category in the response.

The public participation procedures for a remodial action are not forth in § 300.430. Section 300.815(c) of Subpart I required that compliance with the requirements of () 200.420(f) be documented for inclusion in the odznimiotrotivo rocord file. The requirements of \$ 200.420(f) include and booogong of a governed plan: publication of a notice of availability, and brief analysis of the proposed plan: pincing a copy of the proposed plan in the information repository: providing an opportunity for the submission of sal ao danamano ioto to an adiam proposed plaz. RI/FS. and any waiver to elocaup orcadordo wader section 127(d)(d) of CERCLA: providing on opportunity for a public macing on the RI/FS. proposed plaz and waivers to clocaup oroseores: proporing o transcript of public macrings hald during the public comment period: making the transcript available to the public discussing significant changes to the proposed plas: rosponding to significant ලෙකකුලෙහය: ටකල් රම්ස්ස්තුනු පල්ල් පලකුර public common and providing for other public participation precodures at the lead agency's discretion prior to the day word stocky relicion of the religion oubcreasion occidentation of the control of the con will generally be the proctice of the lead ogency that whenever possible. document upon which the relection decinion is becod will be included in the administrative record ale as soon as possible after they are generated or received and no later than when the

decision document to organd. This is unrended to encourage maximum public participation in the development of the record.

Documents generated or received after the selection is saide do not provide a basis for the decision and thus generally are not part of the administrative record. except as provided in § 200.625, discussed below.

5. Administrative record for a removal action (§ 300.820). Section 300.820 proposes requirements for administrative records for removal actions. It is divided into two parts. Paragraph (a) addresses "non-time-critical" removal actions, i.e., those for which, based on the site evaluation, the lead agency determines that a removal action is appropriate and that there is a planning period of at least six months before on-site cleanup activities must be initiated. Paragraph (b) addresses all other removal actions.

Explanations of regulatory requirements and related issues which are the same as those for remedial actions will not be repeated here. Only requirements and issues specific to removal actions will be addressed.

Section 300.820(a)(1) provides that the administrative record file for a non-timecritical removal action shall be available for public inspection when the engineering evaluation/cost analysis (EE/CA) report is made available for public comment. At that time, on administrative record file chall be established and made available to the public and shall contain all documents relevant to selection of the removal action generated up through that date. Documents generally available at that time include sampling data, a preliminary assessment report a site inspection report the EE/CA approval memorandum, and the EE/CA. After the EE/CA report is available and until the Action Memorandum in nigned (except as provided in § 300.823, discussed below). documents relevant to the celection of the removal action shall be added to the administrative record file as discussed in the remodial action section of today's precisable.

The public participation procedured for non-time-critical removal actions are not forth in § 300.419(n)(3) of Subpart E of today's proposed regulations. Section 200.620(a)(3) requires that compliance with § 300.435(n)(3) (i) through (iii) be documented for inclusion in the administrative record. The requirements of § 300.425(a)(3) (i) through (iii) include publication of a notice of availability and brief description of the EE/CA; making the EE/CA available to the public; providing a reasonable opportunity, not less than 30 days, for

submission of comments after the completion of the EE/CA: and responding to semificant comments.

Section 300.820(b) provides different procedures for time-critical, including emergency, removal actions. As explained earlier, section 113(k)(2)(A) of CERCLA requires procedures for the "appropriate" participation of interested persons in the development of the administrative record for removal actions. Appropriate participation is significantly different in situations where an action must be taken on short notice. Where the exigencies of the situation demand that cleanup be initiated and often completed within short umeframes, public comment periods may delay expeditious response to the emergency. In view of Congressional intent that publica participation requirements not itamper or delay emergency removal actions. EPA has considered many options for the appropriate level of public participation. EPA must balance the benefits of public involvement in ndvance of the nelection of a removal action against the need to proceed quickly in emergency situations. EPA believes that the requirements proposed today strike the correct balance.

EPA has had to consider two questions in determining the level of participation for time-critical removals. First, at what point should the administrative record file be made available to the public and occord should there be a formal public comment period on the record? EPA is proposing in (300.820(b)(1) that for all time-critical removals (including emergencies), the record file should be made available to the public no later than 60 days after initiation of on-site removal activity. EPA in choosing to make the record available at this time recognizing that there will be many situations where immediate action must be undertaken to remove threats to human health and the environment before the administrative record file can be assembled and placed in a public docket for inspection. In reviewing npical removal actions. EPA found that generally containment or stabilization (i.e., those activities taken to rotard. reduce, or prevent the opread of a release or threat of release and climinate any immediate threat) at removal cites often are completed within 60 days. Clearly, where circumstances warrant. EPA should focus on addressing the threat at a site. and attend to administrative procedures later. The proposal mean both EPA's charge to protect human health and the environment and the requirement to provide for appropriate public

perticipation, by requiring that the administrative record file be made available to the public no later than 60 days after initiation of removal activities. Making the record available involves: assembling the administrative record file. identifying a publicly accessible location for the record file at or near the cite. finding an acceptable newspaper and placing an advertisement in it to notify the public. and preparing for receipt and evaluation of comments. The proposed requirement that the file be evailable "no later than" 60 days does not preclude making the record file ovailable at an earlier time. if circumstances allow.

EPA is also proposing in \$200.820(b)(2) that the load agency shall, as appropriate, provide a 30-day public comment period to begin at the time the administrative record is made available to the public. Cenerally, when the removal action has not been completed, a public comment period will be considered appropriate at the time the administrative record file is made available to the public. EPA requests comment on whether public comment on whether public comment though be collected as activities that the record is made available.

EPA has also considered other public participation procedures for time-critical removals. They include:

i. Requiring that the record file be made available immediately upon issuing the Action Memorandum and delaying the izitiation of cleanup until after public comment in solicited and responded to. This would allow maximum public participation in selection of the removal, but it is not consistent with the need to provide prompt response for protection of human hoalth and the cavironment at the cite. Such an approach would also be inconsistent with the legislative history which states that administrative procedures established under section 113 chould not be mper concercacy ഭാമാകവ് വല്യായാ

ii. Requiring that the record be made available "promptly" after incuing the Action Messonadus, and then coliciting public comment "ac time allowe." EPA comidend this as a way of addressing the individual nature of removalo. the different time frames that sacy be involved, and the need to provide meaningful opportunities for public comment in cases where time allows. As discussed earlier. EPA pelieves resources abould first be directed toward saitigating threats at a time-critical removal site and that 60 days of ca-site work will allow this. However. EPA is concerned that a

exactered of "prompt" evaluability is too well and would be a course of controversy at each cite. Thus, EPA believes an objective standard in preferable. Similarly, while providing for public comment "as time allows" permits flexibility in the requirements. Catch a rule would require the exercise of judgment and would allow disputes over compliance with this requirement in individual cases. In addition, as discussed above, it is rare that there is enfficient time before beginning a time-critical action to solicit, consider and respond to comments.

in. Delaying the availability of the necord until 120 days after beginning ciesnup and then soliciting public comment. This opproach parallels the community relations requirements (writin 120 days of cleanup for ongoing raspenses. a Community Relations Plan muotibe prepared and an information repository must be made evailable: see § 360.015(a)). This would increase the mumber of a teo at which clean to had been:completed before the public is notified. EPA believes that the increased cleanup time provided under this option generally does not justify the delay in public involvement concerning response selection.

iv. Requiring that the record file be made ovaliatie ofter perioming containment or stanturation at a site where disposal in needed (over 25 percent of removals do not require disposal) and delaying disposal until public comment could be solicited. evaluated and responded to. Taio approach attempts to balance the need for public comment with the urgency of the response, iumiting the response selection undertaken with the benefit of public input to those asp and ofremovals which must be conducted swiftly in order to protect public health and the environment.

There are two major difficulties with this approach. The first concernations between difficulties with this approach. The first concernation with the first concern and "stabilization" in this content and providing indicators to mark their completion. While it is possible, based on experience, to say that the containment or stabilization phase of a removal action is generally completed within 60 days of initiating work. It is much more difficult to determine such completion on a site-opecific basis.

The second difficulty with such a rule is that it fails to take into account several important factors which may make such an approach infeasible in many cases. Specifically, delay of disposal activities may: (a) Create additional unnecessary risks to human health and the environment, and (b) result in needless expenditures of time

and mency. Site conditions, weather conditions, location, public associability, availability of approved disposal facilities, availability of treatment facilities and the effect of the delay on the statutory time and money limitations on removals are only some of the factors to be considered before a site-by-nite determination could be made as to whether or not it was practicable to solicit public comment.

v. Making the record publicly available as in the proposal (i.e., no later than 60 days after initiation of cleanup), but not formally coliciting any public comment. Given the need for quick action on time-critical removals, that they are generally limited in acope, and few cleanup options are feasible, this may be an appropriate approach. This approach, however, would not provide he public with an opportunity for meaningful participation where it might be appropriate in specific removal aituations.

EPA policity comments on the proposed and other considered approaches to public participation on removal actions.

6. Adding documents after selection of response action (§ 300.825). New documents may be added to the record file after the decision document is signed only as provided in § 300.825. Documents generated or received after the decision document (e.g., Action Memorandum or ROD) is signed generally will be kept in a post-decision document file unless and until a determination is made that the document(s) should be placed in the administrative record file, pursuant to § 300.825.

Section 300.825(a) provides that the lead agency may add post-decision documents to the administrative record file in two situations. The limit situation occum when the decision document does not address or reserves a portion of the response action decision. In such cases, the lead agency will continue to add to the administrative record file documents which form the basic for that portion of the decision not addressed or reserved by the decision document. Where appropriate, the lead agency shall provide public notice that the administrative record file for this portion of the decision continues to be available for public impaction and comment. It should be noted that this exception applies to RODs that address as operable unit but leave a portion of the decision on that operable unit open.

The second situation arises when an explanation of significant differences provided for in § 300.435(c) or an amended decision document is required. An explanation of significant differences

is issued when after adoption of a final mandial action plan the remodial oction or enforcement action taken, or the octtlement or consent decree entered into. organizecally differs in ocope performance or cost from the final plan. The record chall include an explanation of ciralican differences and all මුත් අව අවදා වෙන අතර වියුත් අතර මුතුව ලද අතර docinion to modify the response celection decicion. The lead agency shall publish a notice of availability of these documents. On required by occion 117 of CERCLA and as proposed in () 300.435(c). U. in addition, an amended decision document is required, the bobacaro od spulsai llodo bacardo document and all documents that form the booic for the amended decision. The public participation procedures outlined ia Subport E oa cuploactions of oignificant differences and amendments to decision documents shall apply.

Section 300.025(b) provides that the lead agency may, in its discretion, hold additional public comment periods or extend the time for submissions of public comment after the decision document to signed, and may limit such comment to issue for which the lead agency has requested additional comment. This is intended to allow the lead agency to solicit additional comment on the response action whenever it determines that new information or other circumstances warrant additional input.

Section 200.025(c) governs public comments received after the close of the comment period. Under this section, the lead agency will need to consider such comments only if they could not have been submitted during the comment period an invovide entical new informatte trievon: to the responde oelection watch substantially supports the need to oignificantly elter the response action. EPA is proposing the orandard set out in (200.625(c) as providing the best believes perween EPA's desire to remain open to exitical. acw information on the effectiveness of o delected response and the need to make final decinions in order to allow od to sorpressed worthsoms response action. EPA solicity comment on this approach.

D. Compliance With Thin Subpart

An provided in cretion 113(j)(4) of CERCLA, in reviewing alleged procedural error related to the administrative record, a court many disallow cooks or damages only if the errors were so conson and related to mattern of such caseal relevance to the action that the action would have been significantly changed had such errors not been made.

- (C) Socies 200.000(e)(1), (4), (5), and (6) (6) determining the scool for a Fundiance action), (c) (62 permit unrements), and (6) (62 identification ARARs);
- (D) Section 300.405(b), (c), and (d) (on reports of reienses to the NRC):
- (E) Section 300.010 (on removal site evaluation) encopt paragraphs (e)(5) and (0) and the reference to listing releases in CERCUS in (h):
- (F) Section 300.415 (on removal actions) except paragraphs (a)(2), (b)(2)(vii), (b)(4), and (g);
- (C) Section 300.420 (on remedial site evaluation):
- (H) Section 200.430 (on RI/FS and solication of romody) except paragraph (f)(3)(iv)(F): and

(I) Section 200.435 (on RD/RA and operation and maintenance).

- (ii) In addition, other persons undertaking response actions shall provide an opportunity for public comment concerning the selection of the response action. A response action shall not be considered consistent with the NCP unless:
- (A) The person taking the response action compiles with the following NCP provisions regarding public participation, with the exception of e-dministrative record and information isotory requirements stated therein:
 - Section 300.155 (on public mantion and community relations):

(2) Section 300.415(n) (on community relations during removal actions):

- (3) Section 300.430(c) (on community relations during RI/FS) except paragraph (3):
- (4) Section 300.430(f)(1), (2), and (5) (on community relations during selection of remedy); and
- (5) Section 300.435(c) (on community relations during RD/RA and operation and maintenance): or
- (B) The person taking the response action complies with State or local requirements which provide a substantially equivalent opportunity for public involvement in the ebotes of remedy.
- (iii) When selecting the appropriate remedial ocuon, any other person shall, as appropriate, consider the methods of remedying releases listed in Appendix D of this Part.
- (iv) Except for actions taken pursuant to CERCLA sections 104 or 103 or response actions for which reimbursement from the Fund will be sought, any action to be taken by the in agency in §§ 300.415, 300.430, and

is may be taken by the person ing out the response action.

(d) Section 111/ai/21 Cicirus (1)

Forcons, since than those space in paragraphs (d)(1)(i) through (iii) of this

coction, say be able to recive reisburcement of response costs by seems of a claim egainst the Fund. The categories of persons excluded from pursuing this claims authority are:

(i) Federal government:

(ii) State governments, and their political subdivisions, walens they are potentially responsible parties covered by an order or consent decree pursuant to section 122 of CERCLA: and

(iii) Persons operating under a procurement contract or an assistance agreement with the United States with respect to matters covered by that contract or assistance agreement, unless specifically provided therein.

(2) In order to be reliablesed by the Fund, an eligible person must notify the Administrator of EPA or designed prior to taking a response action and receive prior approval, i.e., "proauthorization." for such action.

(3) Presuthorization is E-A a prior approval to submit a waim against the Fund for necessary response costs incurred as a result of carrying out the NCP. All applications for presuthorization will be reviewed to determine whether the request should receive priority for funding. EPA is its discretion, may grant produborization of a claim. Produthorization will be considered only for.

(i) Removal actions pursuant to § 300.415:

- (ii) CERCLA section 104(b) activities.
 (iii) Remedial actions at National
 Priorities List sites pursuant to § 300.435.
- (4) To receive EPA's prior approval. the eligible person must:
- (i) Domonstrate technical and other capabilities to respond activity and effectively to releases of hazardous substances, pollutants, or contaminants; and
- (ii) Establish that the action will be consistent with the NCP in accordance with the elements set forth in paragraphs (c!(3) (i), (ii), and (iii) of this section.
- (5) EPA will grant preauthorization to a ciaim by a party it determines to be potentially liable under section 107 of CERCLA only in accordance with an order issued pursuant to section 103 of CERCLA or a settlement with the Federal government in accordance with section 122 of CERCLA.
- (0) Preauthorization does not establish an enforceable contractual relationship between EPA and the claimant.
- (7) Preauthorization represents EPA's commitment that if funds are appropriated for response actions, the residual action is conducted in accordance with the presumorization decision account it, and usous are reasonable and necessary.

roimburnoment will be made from the Superfund. up to the mercian emount provided in the probuborsation decising document.

- (8) For a claim to be awarded under section 111 of CERCLA. EPA must certify that the cents were necessary and consistent with the produktorization decision decument.
- (a) Section 103(b) potition. Subject to conditions specified in CERCLA section 166(b), any person who has complied with an order issued after October 10. 1823 pursuant to section 163(a) of CERCLA may seek reimbursement for response costs incurred in complying with that order unless the person has waived that right.
- (f) Section 123 minburrement to local governments. Any general purpose unit of local governments. Any general purpose unit of local government that is effected by a release may receive reimburrement for the coops of temporary conferency macoures to prevent or matigate injury to human health or the covernment unbject to the conditions not forth in 40 CFR Part 310. Such reimburrement may not executed \$225.000 for a single response.
- (1) Rolesso from liebility.
 Implementation of response accourse by potentially responsible parties or by any other person does not release those parties from liebility under section 107(a) of CERCLA except as provided in a settlement under section 125 of CERCLA or a Foderal court judgment.

Subpor I—Administrativo Rosord for Solostisa of Roopsaco Assoa

() TOOLEO (LOTESCHERMICH) ON CA

- (a) Conoral requirement. The code agency shall comblish an administrative record that contains the documents that form the basis for the selection of a response action. The lead agency shall compile and maintain the administrative record in accordance with this subpart.
- (b) Administrative records for Federal facilities. (1) If a Federal agency other than EPA is the lead agency for a Federal locality, the Federal agency shall compile and maintain the administrative record for the election of the response action at that facility in accordance with this subpart. EPA may furnish decument which the Federal agency shall place in the administrative record file to chouse that the administrative record includes old documents that form the basis for the selection of the response action.
- (2) EPA or the U.S. Coast Guard shall compile and maintain the administrative record when it is the lead agency of a Federal facility.

- (3) I PA io iavolvod is the colorion of the response action at a Federal facility on the NPL the Federal agency acting as the lead agency obtail provide PA with a copy of the index of documents included in the administrative record file, the RI/FS wortipion, the RI/FS released for public comments received on the RI/FS and proposed plan, any public comments received on the RI/FS and proposed plan, and any other documents EPA may request on a case-by-case basis.
- (c) Administrative record for Stateiesd sites. If a State is the lead agency for a site, the State shall compile and wantian the administrative record for the sciection of the response action at that allo in accordance with this subpart. EPA may require the State to piace additional documents in the ogamicantas secord file to cueme that the administrative record includes all documents which form the boois for the sciection of the response action. The State shall provide EPA with a copy of " the reger of documents included in the odministrative record file. the RI/FS workplan, the RI/FS released for public comment. the proposed plan. any public common: o received on the RI/FS and proposed plan. and any other documents EPA may request on a case-by-case besis.
- (d) Applicability. This subpart applies to all response actions taken under section 104 of CERCLA or sought secured or ordered administratively or judicially under section 103 of CERCLA as follows:
- (1) Remedial actions where the remedial investigation commenced after the promuigation of these regulations: and
- (2) Removal actions where the action memorandum is signed after the promulgation of these regulations.
- (e) For those response actions not included in paragraph (d) of this section, the lead agency shall comply with this subpart to the extent practicable.

The lead agency chall establish a docket at an office of the lead agency or other central location at which documents included in the administrative resord file shall be located and a copy of the documents included in the samminum resord file shall also be made available for public inspection at or near the site of issue, except an provided below:

(a) Sampling and testing data, quality control and quality assurance documentation, and chain of distory for a very not be located at or near the site at 1980e or at the control location.

- provided that the index to the administrative record file indicates the location and availability of this information.
- (b) Cuidance documents not generated specifically for the site of issue need not be located at or near the site at issue need not be located at or near the site at issue, provided that they are maintained at the central location and the index to the administrative record file indicates the location and availability of these guidance documents.
- (c) Publicly available technical literature not generated for the site of issue, each as engineering textbooks, articles from technical journals, and toxicological profiles, need not be located at or near the cite at issue or at the cantral location, provided that the literature is listed in the needs to the admissionary record file or the literature is cited in a document in the record.
- (d) Documento included in the confidencial portion of the administrative record file shall be located only in the control location.
- (e) The communicative record for a removal action where the release or threat of release requires that on-cite removal activities by initiated within Lours of the lead agency's determination that a removal activities easier that a removal activities easier of initiation, acted by available for public imprection only at the central location.

() 360.018 Contonu of the caminatelimity of

- (a) Contents. The administrative record file for celection of a response action typically, but not in all cases, will contain the following types of documents:
- (1) Decuments containing formal information data, and analysis of the factual informetica and date that may form a basia for the aelection of the response action. Such documents may include verified sampling data, quality ccarrol and quality assurance documentation, chain of custody forms. site inspection reports, preliminary assessment and site evaluation reports. ATSOR bealth assessment documents supporting the lead agency 's determination of parameter and substantial endangerment: public health evaluations, and technical and enginering evaluations. In addition, for remedial actions, such documents may include approved work plans. State documentation of applicable or relevant and appropriate requirements, and the RI/FS:
- (2) Guidance documents, technical literature, and site-specific policy

- escentined that earliers a basis for the coloction of the response ection. Such decreases say include guident on conducting remodel investigation and foosibility studies. Guidence of determining applicable or relevant a appropriate requirements. Guidence on rack/exposure assessments. Origineering handbooks. Origineering handbooks. Origineering handbooks. Origineering of a special regulation to a cite. and memorando on off-site disposal capacity:
- (3) Documents received, published, or made available to the public under § 300.645 for remodual actions, or § 300.626 for removal actions. Such documents may include notice of availability of the administrative record file, community relations plan, proposed plan for remodual actions plan, proposed plan for remodual actions, public comments and new information received by the lead agracy, and responses to algorithms to comments:
- (4) Decision documents. Such documents may include action memorando and records of occision:
- (5) Enforcement orders. Such documents may include administrative orders and consent documents and
- (0) An index of the deciments included in the administrative record file.
- (b) Documents not included in V edministrative reserd. The load of is not required to include documents in the administrative record file which do not form a basis for the selection of the response retion. Such decuments include but ore not limited to draft documents, internal memoranda, and day-to-day notes of staff unions such documents centers information that forms the basis of selection of the response action and the information is not included in any other document in the administrative record file.
- (c) Privileged documents. Privileged documents shall not be included in the record file except so provided in paragraph (d) of this section or where such privilege is waived. Privileged documents include but are not limited to documents oubject to the attorney-client attorney work product, deliberative process, or other applicable privilege.
- (d) Confidential file. It information which forms the bosts for the exection of a response ention to include any made of the entire of a response entioning confidential or privileged information and is not otherwise evaluation to the public, the information, to the entent feasible whill be summersed in such a very as made it decided and placed a publicly available portion of the education of the education.

confidence or privileged document much shall be placed in the confidential bortion of the administrative record file. If the information, cannot be a sinformation, cannot be a sized in a disclosable manner, the information shall be placed only in the confidential portion of the administrative record file. All accuments contained in the confidential portion of the administrative record file shall be listed in the index to the file.

\$300.315 Administrative record for a remedial action.

- (a) The administrative record file for the selection of a remedial action shall be made available for public inspection at the commencement of the remedial investigation phase. At such time, the lead agency shall publish in a major local newspaper of general culculation a notice of the availability of the files containing the administrative record.
- (b) The lead agency shall provide a public comment period as specified in § 300.420(f) so that interested persons may submit comments on the selection of the remedial action for inclusion in the administrative record file. The lead acency is not required to respond to comments that were submitted prior to the public comment period. A written response to significant comments succeed during the public comment.
 - all be included in the arative record file.
- (c) The lead agency snall comply with the public participation procedures required in § 300 430(f) and shall document such compliance in the administrative record.
- (d) Documents generated or received after the record of decision is signed shall be added to the administrative record file only as provided in §300.825.

; 300.820 Administrative record for a removal action.

- (a) If, based on the site evaluation, the ead agency determines that a removal action is appropriate and that a planning period of at least six months exists pefore on-site removal activities must be initiated:
- (1) The administrative record file shall be made available for public inspection when the engineering evaluation/cost inalysis (EE/CA) is made available for jublic comment. At such time, the lead igency shall publish in a major local rewspaper of general circulation a otice of the availability of the fue ontaining the administrative record.
- (2) The lead agency shall provide a the first period as specified in 30 that interested persons may ubrasseomments on the selection of the royal action for inclusion in the

- earminismitive record file. The lead agency is not required to respond to comments that were submitted prior to the public comment period. A written response to significant comments submitted during the public comment period shall be included in the administrative record file.
- (3) The lead agency shall comply with the public participation procedures of § 300.415(n) and shall document compliance with § 300.415(n)(3) (i) through (iii) in the administrative record file.
- (4) Documents generated or received after the action memorandum is signed shall be added to the administrative record file only as provided in § 300.825.
- (b) For all removal actions not included in paragraph (a) of this section:
- (1) Documents included in the summistrative record file shall be made; available for public inspection no later than 60 days after initiation of on-site removal activity. At such time, the lead agency shall publish in a major local newspaper of general circulation a notice of availability of the file containing the administrative record.
- (2) The lead agency shall, as appropriate, provide a public comment period of not less than 30 days beginning at the time the administrative record is made available to the public. The lead agency is not required to respond to comments that were submitted prior to the public comment period. A written response to significant comments submitted during the public comment period shall be included in the administrative record file.
- (3) Documents generated or received after the action memorandum is signed shall be added to the administrative record file only as provided in § 300.825.

§ 300.825 Record requirements after decision document is signed.

- (a) The lead agency may add documents to the administrative record file after the decision document selecting the response action has been signed if:
- (1) The documents concern a portion of a response action decision that the decision document does not address or reserves to be decided at a later date; or
- (2) An explanation of significant differences required by § 300.435(c), or an amended decision document is issued, in which case, the explanation of significant differences or amended decision document and all documents that form the basis for the decision to modify the response action shall be aided to the administrative record file.
- (b) The lead agency may hold additional public comment periods or extend the time for the submission of

- REGISTON COMMENT LATER & decision document has been signed on any issues concerning selection of the response action. Such comment shall be limited to the issues for which the lead agency has requested additional comment. All additional comments submitted during such comment periods that are responsive to the request, and any response to these comments, shall be placed in the administrative record file.
- (c) The lead agency is required to consider comments submitted by interested persons after the close of the public comment period only to the extent that the comments contain significant information not contained elsewhere in the administrative record which could not have been submitted during the public comment period and which substantially support the need to significantly after the response action. All such comments and any responses thereto shall be placed in the administrative record file.

Subpert J—Use of Dispersents and Other Chemicals

§ 300.900 General.

- (a) Section 311(c)(2)(G) of the Clean Water Act requires that EPA prepare a schedule of dispersants and other chemicals, if any, that may be used in carrying out the NCP. This subpart makes provisions for such a schedule.
- (b) This subpart applies to the navigable waters of the United States and adjoining shorelines, the waters of the contiguous zone, and the high seas beyond the contiguous zone in connection with activities under the Outer Continental Shelf Lands Act, activities under the Deepwater Port Act of 1974, or activities that may affect natural resources belonging to, appertaining to, or under the exclusive management authority of the United States, including resources under the Magnuson Fishery Conservation and Management Act of 1978.
- (c) This subpart applies to the use of any chemical agents or other additives as defined in Subpart A of this Part that may be used to remove or control oil discharges.

§ 203,805 NCP Product Schodule.

(a) Oil Discharges. (1) EPA shall maintain a schedule of dispersants and other chemical or biological products that may be authorized for use on oil discharges in accordance with the procedures set forth in § 300.910. This schedule, called the NCP Product Schedule, may be obtained from the Emergency Response Division (OS-210), U.S. Environmental Protection Agency.